

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Dengler, Katharina

Working Paper Effectiveness of sequences of classroom training for welfare recipients: What works best in West Germany?

IAB-Discussion Paper, No. 24/2016

Provided in Cooperation with: Institute for Employment Research (IAB)

Suggested Citation: Dengler, Katharina (2016) : Effectiveness of sequences of classroom training for welfare recipients: What works best in West Germany?, IAB-Discussion Paper, No. 24/2016, Institut für Arbeitsmarkt- und Berufsforschung (IAB), Nürnberg

This Version is available at: https://hdl.handle.net/10419/148842

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Institute for Employment Research

The Research Institute of the Federal Employment Agency



IAB-Discussion Paper 24/2016

Articles on labour market issues

Effectiveness of Sequences of Classroom Training for Welfare Recipients

What Works Best in West Germany?

Katharina Dengler

ISSN 2195-2663

Effectiveness of Sequences of Classroom Training for Welfare Recipients

What works best in West Germany?

Katharina Dengler (IAB)

Mit der Reihe "IAB-Discussion Paper" will das Forschungsinstitut der Bundesagentur für Arbeit den Dialog mit der externen Wissenschaft intensivieren. Durch die rasche Verbreitung von Forschungsergebnissen über das Internet soll noch vor Drucklegung Kritik angeregt und Qualität gesichert werden.

The "IAB-Discussion Paper" is published by the research institute of the German Federal Employment Agency in order to intensify the dialogue with the scientific community. The prompt publication of the latest research results via the internet intends to stimulate criticism and to ensure research quality at an early stage before printing.

Contents

Zusammenfassung.41Introduction2Institutional Framework.3Theory3Theory103.13.1Classroom Training10103.2One-Euro-Jobs113.33.3Considered Sequences1244Literature Review.144.1Empirical Evidence on Short-term Training154.24.2Empirical Evidence on One-Euro-Jobs184.34.3Empirical Evidence on Intensified Activation Packages194.44.4Empirical Evidence on Sequences2155Implementation and Method235.11Implementation of Dynamic Window Approach
1Introduction62Institutional Framework83Theory103.1Classroom Training103.2One-Euro-Jobs113.3Considered Sequences124Literature Review144.1Empirical Evidence on Short-term Training154.2Empirical Evidence on One-Euro-Jobs184.3Empirical Evidence on Intensified Activation Packages194.4Empirical Evidence on Sequences215Implementation and Method235.1Implementation of Dynamic Window Approach23
2Institutional Framework.83Theory103.1Classroom Training103.2One-Euro-Jobs113.3Considered Sequences124Literature Review144.1Empirical Evidence on Short-term Training154.2Empirical Evidence on One-Euro-Jobs184.3Empirical Evidence on Intensified Activation Packages194.4Empirical Evidence on Sequences215Implementation and Method235.1Implementation of Dynamic Window Approach23
3 Theory103.1 Classroom Training103.2 One-Euro-Jobs113.3 Considered Sequences124 Literature Review144.1 Empirical Evidence on Short-term Training154.2 Empirical Evidence on One-Euro-Jobs184.3 Empirical Evidence on Intensified Activation Packages194.4 Empirical Evidence on Sequences215 Implementation and Method235.1 Implementation of Dynamic Window Approach23
3.1 Classroom Training103.2 One-Euro-Jobs113.3 Considered Sequences124 Literature Review144.1 Empirical Evidence on Short-term Training154.2 Empirical Evidence on One-Euro-Jobs184.3 Empirical Evidence on Intensified Activation Packages194.4 Empirical Evidence on Sequences215 Implementation and Method235.1 Implementation of Dynamic Window Approach23
3.2 One-Euro-Jobs113.3 Considered Sequences124 Literature Review144.1 Empirical Evidence on Short-term Training154.2 Empirical Evidence on One-Euro-Jobs184.3 Empirical Evidence on Intensified Activation Packages194.4 Empirical Evidence on Sequences215 Implementation and Method235.1 Implementation of Dynamic Window Approach23
3.3 Considered Sequences124 Literature Review144.1 Empirical Evidence on Short-term Training154.2 Empirical Evidence on One-Euro-Jobs184.3 Empirical Evidence on Intensified Activation Packages194.4 Empirical Evidence on Sequences215 Implementation and Method235.1 Implementation of Dynamic Window Approach23
4Literature Review144.1Empirical Evidence on Short-term Training154.2Empirical Evidence on One-Euro-Jobs184.3Empirical Evidence on Intensified Activation Packages194.4Empirical Evidence on Sequences215Implementation and Method235.1Implementation of Dynamic Window Approach23
4.1 Empirical Evidence on Short-term Training
 4.2 Empirical Evidence on One-Euro-Jobs
 4.3 Empirical Evidence on Intensified Activation Packages
 4.4 Empirical Evidence on Sequences
5 Implementation and Method
5.1 Implementation of Dynamic Window Approach23
5.2 Method25
6 Data and Descriptives
6.1 Data and Identification
6.2 Descriptives
7 Results
7.1 Overall Results
7.1.2 Overall Effects 32
7 1 2 1 DATET for (CT CT) versus (LIBILLIBII) 32
7.1.2.2 DATET for (CT,UBII) versus (UBILCT)
7.1.2.3 DATET for (CT.CT) versus (UBII.CT)
7.1.2.4 DATET for (CT,1EJ) versus (UBII,1EJ)
7.2 Heterogeneous Effects
7.3 Robustness Check
8 Summary and Conclusions
References
Figures and Tables
Appendix

Abstract

Sequences of active labour market programmes (ALMPs) may be part of an intensified activation strategy targeting hard-to-place individuals who may be long-term unemployed and who may encounter extreme difficulty in finding jobs. Such sequences are very common among welfare recipients in Germany, but most studies only evaluate either single ALMPs or unemployed individuals' first ALMP. Thus, I analyse the effects of participation in different sequences of classroom training, unemployment benefit II (UB-II)-receipt and One-Euro-Jobs for West German men and women on different labour market outcomes. Using rich administrative data and a dynamic matching approach, I can control for dynamic selection problems that occur during a sequence. My results show that two classroom trainings are more effective than two periods of UB-II-receipt in helping welfare recipients find regular employment, especially among West German women. In some cases, avoiding participation in multiple programmes is preferable: participation in two classroom trainings has mostly no beneficial effects over participation in one classroom training in the second period only and participation in one classroom training followed by a One-Euro-Job has mostly no beneficial effects over participation in a One-Euro-Job in the second period only. Moreover, immediately assigning individuals to classroom training is more effective than waiting and assigning them to classroom training in the second period (the effects of timing) because of the positive effects on avoiding UB-II-receipt (work-test function). However, evidence for programme careers or stepwise integration is only observed for the sequence of two classroom trainings versus the sequence of two periods of UB-II-receipt.

Zusammenfassung

Maßnahmesequenzen können Teil einer intensivierten Aktivierungsstrategie für schwer vermittelbare Personen, die langzeitarbeitslos sind und Schwierigkeiten aufweisen, einen Job zu finden, darstellen. Solche Maßnahmesequenzen sind sehr häufig für ALG-II-Bezieher/-innen in Deutschland beobachtbar, aber bisherige Evaluationen von aktiven Arbeitsmarktprogrammen haben sich meist nur auf eine oder die erste Maßnahme von Arbeitslosen konzentriert. Deswegen werden in dieser Studie die Effekte der Teilnahme an verschiedenen Sequenzen, die aus nichtbetrieblichen Trainingsmaßnahmen, ALG-II-Bezug und Ein-Euro-Jobs bestehen, für Männer und Frauen in Westdeutschland auf verschiedene Arbeitsmarktzielgrößen untersucht. Auf Basis von umfangreichen, administrativen Daten und eines dynamischen Matching Ansatzes, kann für dynamische Selektionsprobleme, die während einer Sequenz auftreten, kontrolliert werden. Die Ergebnisse zeigen, dass zwei aufeinanderfolgende nicht-betriebliche Trainingsmaßnahmen im Vergleich zu zwei Perioden in ALG-II-Bezug ohne Maßnahmeteilnahme positive Effekte auf ungeförderte versicherungspflichtige Beschäftigung aufweisen, insbesondere für westdeutsche Frauen. In manchen Fällen ist eine Vermeidung der Teilnahme an mehrfachen Programmen empfehlenswert: sowohl die Teilnahme an zwei nicht-betrieblichen Trainingsmaßnahmen im Vergleich zur alleinigen Teilnahme an einer nicht-betrieblichen Trainingsmaßnahme in der zweiten Periode als auch die Teilnahme an der Sequenz nicht-betriebliche Trainingsmaßnahme gefolgt von einem Ein-Euro-Job im Vergleich zur alleinigen Teilnahme an einem Ein-Euro-Job in der zweiten Periode weisen meistens keine positiven Effekte auf. Direkt nach Eintritt in den ALG-II-Bezug an einer nicht-betrieblichen Trainingsmaßnahme teilzunehmen ist effektiver als zu warten und erst in einer zweiten Periode an einer nicht-betrieblichen Trainingsmaßnahme teilzunehmen, da positive Effekte, den ALG-II-Bezug zu verlassen, auftreten. Hinweise auf Maßnahmekarrieren oder eine schrittweise Arbeitsmarktintegration ist nur für die Sequenz zwei nicht-betriebliche Trainingsmaßnahmen versus ALG-II-Bezug in zwei Perioden beobachtbar.

JEL classification: C13, I38, J68

Keywords: sequences, dynamic propensity score matching, activation, classroom training, One-Euro-Jobs, West Germany

Acknowledgements: I particularly thank Joachim Wolff, Regina T. Riphahn, Katrin Hohmeyer and Eva Kopf for their very helpful comments. Furthermore, I would like to thank Michael Lechner and his team for very helpful suggestions. All errors are my own.

1 Introduction

Because unemployment rates were high in Germany after German reunification—up to 11.4 percent in 1997 (Department for Statistics of the Federal Employment Agency 2013) – the Hartz reforms were implemented between 2003 and 2005. Afterwards, the unemployment rate decreased to 6.9 percent by 2013 (Department for Statistics of the Federal Employment Agency 2013). However, the share of long-term unemployed per unemployed, defined as being unemployed for at least 12 months, is a major concern in many OECD countries: approximately one-third of the unemployed in 2011 were long-term unemployed (OECD 2013). In Germany, the share of long-term unemployed in above average among OECD countries (Figure 1) and long-term unemployed individuals primarily receive welfare benefits.

Sequences of active labour market programmes (ALMPs) may be more likely among the long-term unemployed, as an individual must be unemployed for an extended period to receive the opportunity to participate in a sequence of ALMPs. Furthermore, sequences of ALMPs may be part of an intensified activation strategy targeting hard-to-place individuals who may be long-term unemployed and who may encounter extreme difficulty in finding jobs. These individuals require more assistance from job centres. Thus, intensified activation packages comprising different phases of activation may help to integrate these individuals stepwise into regular employment. Some countries have implemented such intensified activation packages - some targeting these hard-to-place individuals. An intensified activation strategy could also consist of sequences comprising existing ALMPs, whereby case workers apply a strategic mix of programmes that increasingly aim to integrate hard-to-place individuals into regular employment. However, programme careers may arise, i.e., individuals may participate in many ALMPs for several years with no positive employment effects. Furthermore, such sequences may also be the only way for welfare recipients to become 'employed' (Dengler 2013).

Sequences of ALMPs are very common for German welfare recipients: approximately 40 percent of the individuals participating in a first ALMP also participate in a second ALMP (Dengler/Hohmeyer 2010). Thus, understanding the effectiveness of sequences of ALMPs on labour market outcomes is highly important for policy makers to improve the efficiency of the labour market, to foster welfare recipients' successful and stepwise integration into the labour market and to avoid programme careers. However, most studies evaluate unemployed individuals' first programme only. So far, empirical evidence on sequences of ALMPs for welfare recipients in Germany is scarce.

In this paper, I analyse sequences of classroom training for unemployed welfare recipients in West Germany by using a dynamic matching approach that addresses dynamic selection problems during a sequence. Because this approach is very data hungry, I use rich administrative data from the German Federal Employment Agency. Specifically, I draw an inflow sample of all individuals receiving unemployment

benefit II (UB II) without contributory employment for the period between October 1, 2005, and September 30, 2006.

Because classroom training shows the second-highest inflow between 2005 and 2008 and because classroom training is a very common component of sequences of ALMPs (Dengler/Hohmeyer 2010), I consider possible sequences of classroom training. I analyse four effects on different labour market outcomes. First, I consider a basic comparison similar to the static evaluation of participation in a programme compared with non-participation: i.e., the sequence of two classroom trainings compared with the sequence of two periods of UB-II-receipt. Second, I analyse the effect of timing: i.e., the sequence of classroom training followed by UB-II-receipt compared with the sequence of UB-II-receipt followed by classroom training. Third, I consider the effect of participation in multiple programmes by comparing participation in two programmes with participation in only one programme: i.e., the sequence of two classroom training. Fourth, I analyse the effect of participation in multiple programmes for one very common sequence: the sequence of classroom training followed by a One-Euro-Job compared with the sequence of UB-II-receipt followed by a One-Euro-Job.

My results reveal positive regular employment effects for individuals participating in classroom training in the first period if they participate in two classroom trainings compared with if they receive UB II for two periods, especially for West German women; however, I do not observe similar effects of such a sequence on avoiding UB-II-receipt. Regarding the effect of timing, the results show positive effects on avoiding UB-II-receipt. Thus, comparing the sequence of classroom training followed by UB-II-receipt with the sequence of UB-II-receipt followed by classroom training may test an individual's willingness to work. However, mostly no well-determined effects of participation in multiple programmes emerge. Only some positive effects on leaving UB-II-receipt arise for West German men participating in classroom training in the first period if they participate in the sequence of classroom training followed by a One-Euro-Job versus the sequence of UB-II-receipt followed by a One-Euro-Job. In general, evidence of programme careers of One-Euro-Jobs or stepwise integration with further vocational training is generally not observed (it is observed only for the sequence of two classroom trainings versus the sequence of two periods of UB-II-receipt).

The paper is structured as follows: Section 2 describes the institutional framework of UB II and the considered programmes (i.e., classroom training and One-Euro-Jobs). Section 3 discusses the potential effects of the considered programmes and the considered sequences from a theoretical perspective. Section 4 summarises the literature on the effects of participation in short-term training, One-Euro-Jobs, intensified activation packages and sequences. Section 5 describes the implementation of the dynamic window approach and the dynamic matching approach. Section 6 introduces the data and identification and presents some descriptive statistics. Section 7 presents the results and section 8 concludes.

7

Institutional Framework 2

Germany's high unemployment rates after German reunification led to the introduction of the four so-called Hartz reforms between 2003 and 2005. In 2005, the final reform, called 'Hartz IV', introduced a means-tested benefit, UB II, which is regulated in Social Code (SC) II and is the replacement of the formerly means-tested unemployment assistance and social assistance. Since 2005, a two-tier system has existed with a time-limited unemployment insurance benefit (unemployment benefit I (UB I)) and the means-tested UB II, which needy individuals who are capable of working receive if their other sources of (household) income are insufficient to achieve a minimum standard of living. Thus, not only unemployed individuals who either are not or are no longer entitled to UB I but also UB-I-recipients or even employed people whose household income is insufficient to achieve a minimum standard of living can receive UB II.¹ One important and new element of UB II is its focus on activation for all members of needy households who are capable of working to reduce their dependence on welfare. Thus, several ALMPs were implemented in 2005; in addition, ALMPs that existed before the reforms are now available for UB-IIrecipients or are available in a new design. Since 2005, intensified activation packages such as the JobPerspective in 2007 or the so-called 'Bürgerarbeit' in 2010 have been introduced to integrate very disadvantaged welfare recipients; however, these programmes are not in place anymore. As case workers are very flexible in implementing ALMPs, sequences of ALMPs are a commonly used activation approach in Germany.

As I consider sequences of classroom training and One-Euro-Jobs in the period between 2005 and 2008, I describe the institutional background for this period. Classroom training consists of various programmes with different durations: application training lasts for up to 2 weeks, aptitude tests up to 4 weeks and skill training last for up to 8 weeks. If different types of classroom training are combined, the maximum duration is 12 weeks. The different types of classroom training also have different aims: application training should improve the effectiveness of welfare recipients' job search but can also test welfare recipients' willingness to work. Aptitude tests check welfare recipients' aptitude for a specific occupation or job, whereas skill training involves short-term computer, language, or occupation-specific courses and aims to increase human capital. Classroom training is part of so-called short-term training, which also includes in-firm training. In-firm training is similar to classroom training but is conducted within a company. Welfare recipients do not receive any wages in addition to their UB-II-benefits during short-term training; however, job centres pay additional costs that arise out of programme participation, such as childcare or travel costs.²

In my paper, I consider welfare recipients who are unemployed.

See Kopf (2013) and Wolff/Jozwiak (2007).

Short-term training was introduced in 1998 with the SC III (Article 48-52), and it is available to UB-II-recipients via Article 16 of SC II.³ In 2009, the German government reorganised the ALMPs on an evaluation-based level. The reform abolished inefficient ALMPs and modified existing ALMPs to increase their transparency and efficiency.⁴ Since 2009, short-term training has been reorganised into various training modules at an individual or team level, and it is regulated in Article 46 of SC III as an element of 'activation and occupational integration schemes'. In 2012, the German government continued the evaluation-based reforms of ALMPs with legislation on improving the integration opportunities on the labour market by reorganising existing instruments.⁵ The reform aimed to increase decentralisation, flexibility, transparency and individual needs. Since 2012, short-term training has been regulated in Article 45 of SC III.

One-Euro-Jobs have been introduced via the enforcement of SC II (regulated in Article 16d of SC II) and are a public employment programme. One-Euro-Jobs ('Arbeitsgelegenheiten in der Mehraufwandsvariante') constitute work opportunities that provide additional jobs that would not be made available without a subsidy, that have a public interest and that pay 1-2 Euros per hour in addition to welfare benefits. The average duration of a One-Euro-Job is up to 6 months and the average working time is no more than 30 hours per week (Department for Statistics of the Federal Employment Agency 2006-2009). One-Euro-Jobs should focus on hard-to-place welfare recipients, such as uneducated individuals, older individuals or individuals with migration backgrounds. Furthermore, young adults are a special target group of the SC II because job centres have to place them without delay into employment, vocational training, or, as a last resort, a One-Euro-Job.⁶ One-Euro-Jobs may have various purposes for different types of participants: First, One-Euro-Jobs should increase welfare recipients' employability. Thus, the primary goal of One-Euro-Jobs is not welfare recipients' integration into regular employment. Second, One-Euro-Jobs can also test welfare recipients' willingness to work.⁷

Furthermore, I consider the inflow of West German men and women into major ALMPs for the period between 2005 and 2008 (Table 1).⁸ One-Euro-Jobs are the most frequent ALMP: between 100,000 and 300,000 West German participants entered a One-Euro-Job each year. However, classroom training is the second most common ALMP: the annual inflow into classroom training amounts to approximately

⁵ See Bellmann et al. (2011).

⁷ See Dengler (2013).

9

³ See Wolff/Jozwiak (2007).

⁴ See Steinke et al. (2012).

⁶ Since April 2012, job centres are no longer required to place young adults in One-Euro-Jobs without delay.

⁸ Figures on classroom training after the 2009 reform cannot be provided because classroom training is only one element of the new 'activation and occupational integration scheme' (Article 46, SC III) that summarises various ALMPs (e.g., classroom training, private placement services and personal service agencies).

100,000 for West German men and ranges from approximately 70,000 to 90,000 for West German women. Inflow into One-Euro-Jobs and classroom training is higher for West German men than for West German women. However, the average stock of unemployed UB-II-recipients is also higher for West German men.

3 Theory

In this section, I discuss not only the potential effects of the considered ALMPs (classroom training and One-Euro-Jobs) but also the potential effects of the considered sequences. In general, a priori the effects are unclear. Single ALMPs may have both positive and negative effects on wages and employment according to the matching theory (Pissarides 1979). Sequences of ALMPs may strengthen both positive and negative effects.

3.1 Classroom Training

I discuss the potential effects of classroom training in a matching theory framework that implies both negative and positive effects of classroom training on employment (Calmfors 1994; Hagen/Steiner 2000).

Classroom training may have an effect on employment by increasing the efficiency of the matching process:⁹ First, classroom training enhances participants' human capital. Classroom training such as skill training or aptitude tests may increase either general or job-specific human capital. Thus, classroom training improves matching quality because of participants' qualifications become better adapted to the labour demand. Second, classroom training supports and promotes job search activities among participants. Furthermore, classroom training such as application training can test participants' willingness to work. Thus, obligatory programme participation may lead to an increase in job search intensity as participants' free time declines and as earning additional money through illegal employment becomes impossible. Additionally, classroom training provides participants with information such as information about the regional labour market helping participants in their job search. In the same sense, aptitude tests provide participants with information by testing their aptitude to work in a specific occupation. Thus, classroom training can accelerate the matching process through better and higher job search activities. Third, classroom training may send positive signals to potential employers through certificates. Thus, classroom training facilitates the matching process through a screening function.

However, classroom training may also have negative employment effects. For instance, lock-in effects may arise because the participants have less time to search for a job. If individuals know that they will participate in a programme in advance, their job search effort may also decline prior to their participation in the programme (Ashenfelter's Dip). Stigma effects can also emerge if potential employers regard

⁹ See also Kopf (2013) for a discussion of the potential effects of short-term training.

classroom training as a negative signal. Furthermore, individuals may be assigned to classroom training when they need a different programme (wrong allocation) or no programme at all because they already have good employment prospects (creaming). Both creaming and wrong allocation may have long-lasting negative employment effects. Financial disincentives may also occur if job centres pay additional costs that arise out of programme participation.

Because classroom training consists of short courses, classroom training may prepare welfare recipients for other programmes to integrate them stepwise into regular employment. For example, classroom training may provide participants with basic or specific skills in short courses to prepare them for further vocational training that offers specific professional skills or a vocational training degree. Furthermore, classroom training may not subject participants to programme careers because classroom training consists of short courses only and because it cannot 'employ' welfare recipients for a longer time period, as with One-Euro-Jobs.

In summary, a priori the potential effects of classroom training are unclear. Because classroom training occurs over short periods of up to 12 weeks, considerable lock-in effects are unlikely. Thus, I expect classroom training to have positive employment effects.

3.2 One-Euro-Jobs

In this section, I discuss the potential effects of One-Euro-Jobs again in a matching theory framework that implies both negative and positive effects of One-Euro-Jobs on employment (Calmfors 1994; Hagen/Steiner 2000).

Similar to short-term training, One-Euro-Jobs may have an effect on employment by increasing matching efficiency: First, participants in One-Euro-Jobs may become accustomed to regular work routines, are trained on the job and may experience an increase in their human capital because some One-Euro-Jobs may also include a qualification component. Second, One-Euro-Jobs may increase participants' job search activities, as they may test participants' willingness to work. Thus, obligatory participation may increase the intensity of job search as participants' free time declines and as earning money by illegal employment becomes impossible. Third, One-Euro-Jobs may provide positive signals to potential employers.

One-Euro-Jobs may also have negative employment effects, however. For instance, lock-in effects, Ashenfelter's dip and stigma effects may emerge. Because the average duration of One-Euro-Jobs is up to 6 months (Department for Statistics of the Federal Employment Agency 2006-2009), lock-in effects may be substantial. However, the average working time in a One-Euro-Job is no more than 30 hours per week (Department for Statistics of the Federal Employment Agency 2006-2009). Thus, sufficient time to search for a job may remain available. Moreover, because One-Euro-Jobs are generally held by hard-to-place individuals, stigma effects may

emerge. Creaming, wrong allocation and financial disincentives may also arise from One-Euro-Job participation.

Because One-Euro-Jobs are generally held by hard-to-place individuals who have very low chances of finding a job, One-Euro-Jobs need to only increase welfare recipients' employability and chances of finding regular employment; the primary goal does not need to be the integration of welfare recipients into regular employment. Thus, One-Euro-Jobs may be only a first step to prepare welfare recipients for other programmes that then aim to integrate welfare recipients into regular employment. However, One-Euro-Jobs might lead to programme careers.¹⁰

In summary, the potential effects of One-Euro-Jobs are a priori unclear. I expect to find only some small positive or negative employment effects for One-Euro-Job participation, as One-Euro-Jobs' primary goal is not integration into regular employment.

3.3 Considered Sequences

Sequences of ALMPs may arise for two reasons: First, such sequences may arise as part of a stepwise and intensified activation strategy. In particular, hard-to-place individuals may need more than one programme participation to find regular employment. Thus, case workers may integrate individuals stepwise into regular employment by employing a strategic mix of programmes. Second, sequences of ALMPs may also be part of a programme career as the only way for welfare recipients to be 'employed'.

In general, sequences of ALMPs may strengthen both the positive and the negative employment effects of single ALMPs. First, I discuss the potential effects of the sequence of two consecutive classroom trainings. Two consecutive classroom trainings may increase human capital because the participants may acquire more gualifications and skills. In particular, if the two classroom trainings build on one another and/or if a case worker assigns welfare recipients to two different types of classroom training (e.g., initial basic skill training followed by more specific skill training or initial basic application training followed by skill training), the positive effects of classroom training on employment may be increased. An aptitude test as a first programme followed by specific skill training may also be beneficial, as the case worker may first test the participant's aptitude in a specific occupation and skill training may then provide missing skills. Furthermore, two classroom trainings may enhance participants' job search activities, provide participants with more information and send more positive signals (e.g., two certificates) to potential employers. However, lock-in effects or stigma effects that are normally not a major concern in the case of classroom training may arise because of the longer duration of the programme participation. Because each of the two classroom trainings remains very short, I do not ex-

¹⁰ See Dengler (2013).

pect substantial lock-in or stigma effects. Thus, I expect a sequence of two consecutive classroom trainings to have positive effects on employment through higher accumulation of human capital and job search activities.

Second, I consider the potential effects of the sequence of UB-II-receipt followed by classroom training. On the one hand, individuals who wait for their first classroom training may obtain classroom training consistent with their preferences, former tasks and skills. Furthermore, such individuals may have sufficient time to search for a job (no lock-in effects) leading to a faster integration into regular employment, as they do not have to take part immediately in classroom training. On the other hand, the loss of human capital and efforts to integrate individuals into employment (matching efforts) may be higher because of their longer duration of unemployment (without programme participation). Furthermore, individuals who wait for their first classroom training cannot quickly leave UB-II-receipt and stigma effects arising from unemployment without programme participation may emerge. In summary, I expect negative effects dominate in the sequence of UB-II-receipt followed by classroom training because classroom training is a short programme that is cheap and that may be offered frequently; thus, it may be not necessary for individuals to wait to receive their first classroom training to obtain appropriate classroom training. Furthermore, considerable lock-in effects are unlikely for classroom training; thus, individuals that immediately receive classroom training may also have sufficient time to search for a job.

Third, I consider the potential effects of the sequence of classroom training followed by UB-II-receipt. Individuals who immediately receive classroom training after entry into UB-II-receipt may not experience a great loss of human capital, suffer from stigma effects or exert high matching efforts because they would avoid unemployment without programme participation. Furthermore, integration into regular employment may be faster for such individuals than for individuals who wait to receive their first classroom training. However, individuals who immediately receive classroom training are less likely to obtain appropriate classroom training and such individuals may not have sufficient time to search for a job immediately (lock-in effects). Because classroom training also tests individuals' willingness to work, individuals who immediately receive classroom training may leave UB-II-receipt more quickly. However, the work test function might lead to only the avoidance of UB-II-receipt, not necessarily positive regular employment effects. In summary, I expect to find positive effects of this sequence on employment because the diminished appropriateness of classroom training and potential lock-in effects may have only a small negative effect.

Fourth, I describe the potential effects of the sequence of classroom training followed by a One-Euro-Job. Beneficial effects of such a sequence may arise if individuals first obtain specific skill training and then become accustomed to basic work schedules – or if they train on the job in a consecutive One-Euro-Job. Thus, the sequence of classroom training followed by a One-Euro-Job may have positive effects on employment. However, classroom training as first programme may also screen the participants' skills and productivity: case workers assign individuals who show low productivity in classroom training to a consecutive One-Euro-Job. As One-Euro-Jobs are a last resort, individuals may be assigned to One-Euro-Jobs after classroom training as the only alternative for being 'employed'. Furthermore, lock-in effects and especially stigma effects may be a notable concern with One-Euro-Jobs. Thus, in summary, negative effects on employment dominate for this sequence.

Fifth, I consider the potential effects of the sequence of UB-II-receipt followed by One-Euro-Job participation. Because individuals who wait for their first One-Euro-Job are more likely to receive an appropriate One-Euro-Job – e.g., a One-Euro-Job consistent with their preferences for specific tasks and former job skills – positive employment effects may arise. Furthermore, such individuals may have sufficient time to search for a job (no lock-in effects) leading to a faster integration into regular employment, as they do not have to take part immediately in a One-Euro-Job. However, for individuals who wait for their first One-Euro-Job, negative employment effects may arise because the loss of human capital, stigma effects and matching efforts may be higher due to the longer duration of unemployment (without programme participation). Furthermore, lock-in effects and especially stigma effects may be a notable concern with One-Euro-Jobs. In summary, I expect small positive effects on employment as the appropriateness of a One-Euro-Job may play a major role as various types of One-Euro-Jobs in different sectors exist.

Sixth, I consider the potential effects of the sequence of two periods of UB-II-receipt without programme participation, i.e., individuals receive only welfare benefits. Negative employment effects due to the loss of human capital, stigma effects and higher matching efforts may arise because of the high unemployment duration without programme participation. However, such individuals have sufficient time to search for a job as they are not locked in a programme. As lock-in effects may be also not considerable during classroom training and One-Euro-Jobs, but classroom training or One-Euro-Jobs provide at least, e.g., skills, training on the job or accustoming to basic work schedules, I expect negative effects of this sequence on employment.

4 Literature Review

First, I review the empirical literature for evidence on short-term training for Germany and other countries. For Germany, I provide a short overview on the effectiveness of short-term training for unemployment insurance or unemployment assistance recipients before 2005. As I analyse classroom training for UB-II-recipients in Germany, the most relevant literature review relates to short-term training for welfare recipients after the introduction of UB II in 2005. Second, I provide a short overview on the effects of participation in public employment programmes for other countries and a broad overview on the effects of participation in One-Euro-Jobs for Germany. Third, I review the international literature on the effectiveness of intensified activation packages. Fourth, I summarise the literature on the effects of participation in sequences of ALMPs for Germany and other countries.

4.1 Empirical Evidence on Short-term Training

Training programmes are a very common type of ALMPs in many countries and different types of training programmes exist, including not only short-term training (classroom training and in-firm training) but also (further) vocational training or retraining to earn a vocational degree in a different occupation. Immervoll/Scarpetta (2012) show that in most OECD countries, training programmes compose the largest spending category in terms of overall spending on ALMPs. Several metaanalyses - e.g., Card/Kluve/Weber (2010) for various countries and ALMPs, Kluve (2010) for European countries and various ALMPs, Greenberg/Michalopoulos/Robins (2003) for the US and training programmes summarise evaluation studies. For example, Card/Kluve/Weber (2010) reveal positive impacts for participation in short-term training in the medium run (measured approximately 2 years after programme completion), but negative impacts in the short run (measured approximately 1 year after programme completion).

For Germany, many studies analyse the effects of participation in short-term training for unemployment insurance recipients or unemployment assistance recipients prior to the Hartz IV reforms. Most of these studies use matching methods. Analysing short-term training in September 2002 on the avoidance of unemployment, Stephan/Rässler/Schewe (2006) find positive effects of participation in in-firm training for East Germany and negative or not well-determined effects for participation in classroom training 2 years after the start of the programme. Biewen et al. (2007) analyse Germany's short-term training in the early 2000s and their results indicate that such training has positive regular employment effects for West German men and women within 2 to 2.5 years after programme start following a short lock-in period of approximately 3 months, especially for those who began the programme later in their unemployment spell. Büttner (2008) analyses short-term training that tests individuals' willingness to work (part of the application training) by using experimental data for 2005. His results reveal some positive announcement effects (threat effects) and programme effects on avoiding unemployment and on unsubsidised employment up to 1 year after programme start, whereas the threat effects are higher. Stephan (2008) analyses short-term training starting in March 2003 and finds positive regular employment effects of participation in in-firm training 3.5 years after programme start but only some positive effects of participation in classroom training. Furthermore, the author finds evidence of programme careers. Wunsch/Lechner (2008) find no positive regular employment effects of participation in short-term training in the early 2000s within 2.5 years after programme start for West Germany (only some positive regular employment effects for the subgroup of participants with no vocational degree approximately 12 months after programme start). Stephan/Pahnke (2011) also investigate short-term training in Germany starting in March 2003 and find positive regular employment effects of participation in in-firm training and in classroom training within 3.5 years after programme start. For the periods between 1980 and 1992 (old short-term training) and between 2000 and 2003 (new short-term training), Fitzenberger et al. (2013) analyse the effects of participation in short-term training on employment, earnings, and participation in longterm training programmes in West Germany. Their results for the new short-term training show positive regular employment effects after a short lock-in period (1 to 4 months) within 3 years after programme start and future participation in long-term training programmes that indicates stepwise integration with the combination of short-term training and long-term training.¹¹ Furthermore, two studies use a duration model framework: Hujer/Thomsen/Zeiss (2006) find positive effects of participation in short-term training in West Germany before 2005 because of a reduction in unemployment duration. Osikominu (2013) finds that short-term training reduces unemployment duration and increases employment stability for West German participants in the early 2000s.

Recently, an increasingly number of studies have analysed short-term training for welfare recipients in Germany after the introduction of the Hartz IV reforms that implemented short-term training for welfare recipients.¹² All these studies, except for Zabel (2013), use matching methods. Wolff/Jozwiak (2007) evaluate classroom training and in-firm training starting immediately after the Hartz IV reforms by using administrative data. They find nearly no lock-in effects and positive regular employment effects of participation in classroom training (approximately 3 to 4 percentage points 20 months after programme start) and in in-firm training (approximately 13 to 22 percentage points 20 months after programme start). Using a stock sample of welfare recipients in October 2006 and programme starts between October 2006 and March 2007, Huber et al. (2011) find positive insured employment effects¹³ of participation in short-term training (approximately 9 percentage points from 7 to 17 months after programme start) based on survey and administrative data. Kopf (2013) analyses various types of short-term training based on the data of Wolff/Jozwiak (2007). Her results indicate that in-firm training has positive regular employment effects (approximately 13 to 20 percentage points 2 to 28 months after programme start) and that classroom skill training (approximately 2 to 5 percentage points 3 to 28 months after programme start), classroom aptitude tests (approximately 2 to 4 percentage points 6 to 28 months after programme start) and combinations of classroom training (approximately 2 percentage points 18 to 28 months only for West German men) also have some positive regular employment effects. However, her results show that classroom application training is rather ineffective (no well-determined or negative regular employment effects up to 28 months after the programme start).

¹¹ Old short-term training shows not only positive and significant effects on employment (subsidised and unsubsidised) but also positive effects on future participation in long-term training.

¹² Because I consider welfare recipients in my paper, I provide only the size of the effects for the studies that also analyse welfare recipients.

¹³ The authors define insured employment as regular employment that is subject to social contributions. It is not clear if only unsubsidised employment is considered.

Various studies also analyse short-term training for different subgroups. For example, Wolff/Jozwiak (2007) also analyse classroom training and in-firm training for different subgroups, such as age groups. They find that classroom training is less effective for young adults aged 18 to 24 years: only men show some positive regular employment effects (approximately 2 percentage points 6 months after programme start for East German men and 5 percentage points 20 months after programme start for West German men). Further, their results show that although in-firm training effective for all age groups, its effectiveness increases with age. is Hartig/Jozwiak/Wolff (2008) analyse short-term training for younger welfare recipients (aged 15 to 25 years) and find positive regular employment effects of participation in in-firm training (approximately 11 to 20 percentage points 12 months after programme start); however, the regular employment effects of participation in classroom training are mostly not well-determined. By contrast, analysing the effects of participation in short-term training on older German welfare recipients (aged 50 years or older) at the end of 2005, Romeu/Wolff (2011) find that classroom training has positive regular employment effects for West German men (approximately 2 percentage points 12 months after programme start) and some positive regular employment effects for East German women 3 to 8 months after programme start (approximately 1 percentage point) but find that in-firm training is guite effective in terms of regular employment (approximately 10 to more than 16 percentage points 3 to 21 months after programme start). Achatz et al. (2012) analyse One-Euro-Jobs, classroom training and in-firm training for young welfare recipients (aged 18 to 30 years) at the end of 2005 and find that in-firm training has positive regular employment effects (ranging from approximately 9 to more than 20 percentage points) 30 months after programme start. However, their results show that classroom training has positive effects 30 months after programme start only for some subgroups, such as single men or women without children (less than 5 percentage points). Examining short-term training male immigrants for and natives. Thomsen/Walter/Aldashev (2013) find mixed effects for different types of short-term training on the drop-off rate from welfare conditional on the take-up of contributory employment for men: aptitude training has positive employment effects for natives and immigrants (between approximately 4 to more than 15 percentage points during the first year after programme start), whereas skill training has positive employment effects only if it is applied early in the welfare spell (approximately 6 percentage points for immigrants and approximately 10 percentage points for natives 1 year after programme start). However, their results show that application training and combined short-term training are ineffective. Zabel (2013) evaluates One-Euro-Jobs and training programmes for single mothers by using a timing-of-events approach and her results regarding classroom training reveal positive regular employment effects for some groups of single mothers in West Germany.¹⁴



The effects relate to relative entry rates, with values above 1 indicating positive effects and values below 1 indicating negative effects. For example, entry rates into regular em-

4.2 Empirical Evidence on One-Euro-Jobs

Several studies analyse public employment programmes for different countries. Card/Kluve/Weber (2010) provide meta-analyses for various countries and find that public employment programmes are a less effective programme compared with other ALMPs in the short run (approximately 1 year after programme completion) and in the medium run (approximately 2 years after programme completion). The study of Kluve (2010) that provides a meta-analysis only for European countries also reveals that public sector employments are less likely to have positive impacts. However, evidence on workfare programmes such as the German One-Euro-Jobs is scarce. For example, Dahl (2003) analyse a workfare programme on earnings and employment in Norway for social assistance recipients, but he does not find significant effects.

For Germany, several studies analyse the effectiveness of single One-Euro-Jobs by using propensity score matching. I provide the most relevant studies in the following. Hohmeyer/Wolff (2010) analyse the effects of participation in different public employment programmes (traditional job creation schemes, work opportunities as contributory employment and One-Euro-Jobs) by using a stock sample of welfare recipients in April 2005. Their results reveal negative effects of One-Euro-Jobs compared with non-participation on regular employment during the first months after the start of the programme, but positive effects on regular employment arise for East German women and West German men and women 3 years after the start of the programme (e.g., approximately 3 percentage points for West German women). Huber et al. (2011) evaluate One-Euro-Jobs between October 2006 and March 2007 by using survey and administrative data. They find positive effects on insured employment15 of participation in One-Euro-Jobs for men, for individuals who are not lone parents and for individuals who have no migration background (approximately 7 to 9 percentage points between 7 and 17 months after the programme start). Hohmeyer (2012) investigates the effects of participation in different types of One-Euro-Jobs by planned duration and weekly working hours compared with non-participation by using a stock sample of welfare recipients of January 2005. Her results reveal positive effects on regular employment in the medium run (e.g., approximately 3 percentage points 28 months after the start of the programme for West German women), but not for East German men. More intensive One-Euro-Jobs have negative effects on regular employment in the medium run for East German men (e.g., approximately -2 percentage points 28 months after the start of the programme for One-Euro-Jobs with planned durations of 8 to 12 months compared with nonparticipation). However, the most positive effects are found for East German women with a medium level of working hours between 21 and 29 hours compared with non-

ployment are raised by a factor of 1.51 for single mothers with a youngest child aged 3 to 5 years in West Germany 12 months after programme start.

¹⁵ The authors define insured employment as regular employment that is subject to social contributions. It is not clear if only unsubsidised employment is considered.

participation (approximately 3 percentage points 28 months after the start of the programme). For West Germany, participation with a planned duration longer than 4 months increases employment prospects. Hohmeyer/Wolff (2012) analyse One-Euro-Jobs versus non-participation by using the same stock sample as Hohmeyer (2012). They find that One-Euro-Jobs have small lock-in effects during the first months after the start of the programme, but positive effects on regular employment arise for women (approximately 3 percentage points 20 months after the start of the programme points 20 months after the start of the programme).

Furthermore, some studies analyse One-Euro-Jobs for specific subgroups. Hohmeyer/Wolff (2012) also analyse One-Euro-Jobs for different age groups and specific disadvantaged groups. Their results reveal negative and not welldetermined effects on regular employment for young welfare recipients (aged 15 to 24 years) 20 months after programme start. Moreover, One-Euro-Jobs are quite effective for individuals who are jobless for longer periods. Thomsen/Walter (2010) analyse the effects of participation in One-Euro-Jobs compared with nonparticipation among immigrants and natives by using an inflow sample of welfare recipients in 2006. Their results reveal negative effects on regular employment conditional on leaving welfare receipt (approximately 3 percentage points for immigrants and natives 1 year after the start of the programme). Wolff/Popp/Zabel (2010) analyse One-Euro-Jobs for welfare recipients aged 15 to 24 years for different qualification and work experience levels. They find no or even negative effects on regular employment and on the avoidance of UB-II-receipt, but some positive effects for young welfare recipients with no qualifications or low work experience. Achatz et al. (2012) evaluate One-Euro-Jobs for young welfare recipients (aged 18 to 30 years). Their results reveal negative effects on regular employment 30 months after the start of the programme, but positive effects for women with a partner and children. Zabel (2013) analyses One-Euro-Jobs for single mothers and finds positive regular employment effects, but not for West German women with children aged 3 to 5 years.

4.3 Empirical Evidence on Intensified Activation Packages

Some countries have implemented intensified activation packages comprising different phases of activation. I will provide some selected examples for Denmark, Germany, Great Britain and Norway. In Denmark, a social experiment with intensified activation has been introduced (Graversen/van Ours 2008). Approximately half of the unemployment insurance recipients who became unemployed between November 2005 and February 2006 were assigned to an intensified activation programme, while the other half only receives activation as usual. After 5 to 6 weeks of unemployment individuals have to participate in a two-week job search programme. Thereafter, individuals have meetings with a case worker every week or every second week and after 4 months of unemployment individuals have to participate in an activation programme with duration of at least 3 months. Graversen/van Ours (2008) analyse the effects of participation in the intensified activation programme on the job finding rate¹⁶ by directly comparing the job finding rates that are defined by mixed proportional hazard specifications of the treatment and control group. The authors find an on average higher job finding rate by 30 percent for the treatment group compared with the control group. Rosholm (2008) also analyse this social experiment, but in a more detailed manner by using a duration model approach to identify separate effects of each programme and to consider dynamic selection bias. He finds a high effectiveness of the intensified activation programme: the exit rate from unemployment ranges between 20 to 40 percent approximately 10 to 26 weeks after entry into unemployment. However, none of the single programmes have a positive effect on the exit rate from unemployment, but the risk of participation in a programme has a strong positive effect on the exit rate. Thus, a reason for the positive effect of the overall intensified activation programme could be threat Pedersen/Rosholm/Svarer (2012) analyse the effects of a similar effects. randomised experiment in Denmark implemented in 2008. The treatment group receives not an overall intensified activation programme, but an intensified treatment consisting of one out of four single elements-weekly group meetings of unemployed individuals with one or two case workers, individual meetings with case worker every 2 weeks, participation in an ALMP after approximately 3 months of unemployment or combination of group meetings and participation in an ALMP. However, the control group receives treatment as usual (meeting with a caseworker at least every 3 months and partici-pation in an activation programme after 9 months of unemployment). The results reveal that individual meetings increase employment¹⁷ by 5 weeks 2 years after the start of the experiment. For men, the effects of participation in the activation programme are positive and significant (approximately 5 weeks 2 years after the start of the experiment) that already arise before the start of the activation programme (threat effects).

In Germany, the so-called JobPerspective was introduced in 2007 for hard-to-place welfare recipients who have been long-term unemployed with at least two additional severe employment impediments. The JobPerspective provides a non-temporary wage subsidy to employers who hire these individuals. However, these individuals have to participate in an intensified activation phase for at least 6 months comprising existing ALMPs. They receive the wage subsidy only if they still do not find regular employment after they complete the activation phase. However, no study analyses the combination of the activation phase and the wage subsidy. Dengler et al. (2013) analyse only the implementation of the activation phase and their impact on the labour market by using administrative data and a difference-in-difference approach. The authors find a modestly intensified activation for the treatment group leading to subsidised employments other than the JobPerspective, especially in East Germany (e.g., treatment group members spend approximately 0.2 to 0.6 days more in subsidised employments).

¹⁶ The job finding rate is defined as transitions out of the benefit system into employment.

¹⁷ Employment is not defined in a more detailed manner in the study.

In Great Britain, the (flexible) New Deals provide intensive employment assistance comprising a personal advisor and different programmes at certain points of time for different target groups of unemployed individuals (Finn/Schulte 2008). Several evaluation studies analyse the New Deals. For example, Lissenburgh (2004) analyse the relative effectiveness of various options of the New Deal for Young People (NDYP) between 1998 and 1999 in Scotland. The NDYP consists of the so-called Gateway period where individuals receive intensive help and counselling. If individuals still do not get a regular employment, they have to participate in an activation programme (subsidised employment in the private sector, full-time education, work for the voluntary sector or work with the environment task force). By using a matching approach, his results show that the subsidised employment option is the most effective one (e.g., positive effects to leave unemployment arise by approximately 7.7 percentage points 20 to 24 months after programme start if individuals participate in the subsidised employment option compared with the full-time education option).

In Norway, Rønsen/Skarðhamar (2009) analyse a comprehensive action plan for welfare recipients introduced in 2003. The programme consists of several rehabilitation and activation programmes and a strong cooperation between national and local welfare systems. By using administrative data and survival analysis, their results reveal positive effects on employment (unsubsidised and subsidised) for long-term unemployed welfare recipients: participants show a higher average employment entry rate by 37 percent compared with non-participants over the 20 months after entry into the programme.

4.4 Empirical Evidence on Sequences

In general, evidence on the effects of participation in sequences of ALMPs is scarce. For Germany, three studies analyse sequences of ALMPs. Jaenichen/Stephan (2011) analyse the sequence of in-firm training followed by a wage subsidy (paid to employers) versus only in-firm training for unemployed hard-to-place individuals in the early 2000s by using a static matching approach that does not consider intermediate outcomes. Their results show positive regular employment effects (approximately 14 to 31 percentage points 3 years after the start of the wage subsidy). Using a dynamic matching approach, Lechner/Miguel (2010) analyse training programmes for unemployment insurance and assistance benefit recipients in West Germany. They do not consider short-term training but instead the training programmes in the early 1990s, namely, vocational training programmes (T) and retraining (R). They evaluate the employment (subsidised and regular employment) effects of spending four guarters in the considered programmes or being unemployed (U): TTTT versus RRRR, TTTT versus UUUU and RRRR versus UUUU. Their results reveal positive employment effects for the sequence of four quarters of retraining versus the sequence of four quarters of unemployment (approximately 35 percentage points 4 years after entry into unemployment). In addition, regarding the comparison between the sequence of retraining and the sequence of vocational training programmes, vocational training programmes lead to faster integration into employment than retraining (approximately -26 percentage points 2 years after entry into unemployment). Dengler (2013) analyses sequences of One-Euro-Jobs by using an inflow sample of welfare recipients between October 2005 and September 2006 and a dynamic matching approach. She applies a dynamic window approach for the definition of sequences and compares the sequence of two consecutive One-Euro-Jobs with the sequence of two periods in UB-II-receipt and the sequence of a One-Euro-Job followed by UB-II-receipt with the sequence of UB-II-receipt followed by a One-Euro-Job. Her results reveal positive effects on regular employment for female participants in One-Euro-Jobs in the first period, especially in West Germany, if they participate in two One-Euro-Jobs compared with receiving UB-II-receipt for two periods (approximately 12 percentage points 13 months after the start of the second programme). For male participants in One-Euro-Jobs in the first period, especially in East Germany, positive effects on regular employment arise if they participate immediately in a One-Euro-Job versus waiting for a One-Euro-Job in the second period (approximately up to 6 percentage points 13 months after the start of the second programme). Positive effects on regular employment also arise for East German women (approximately 3 percentage points 12 months after the start of the second programme).

Some studies examine the effectiveness of sequences of ALMPs in other countries. Using a dynamic matching approach, Lechner (2004) evaluates sequences of ALMPs for unemployed individuals in Switzerland in the late 1990s. He considers four different states: unemployment (U), training courses (C), employment programmes (E) and temporary wage subsidies (T).¹⁸ One period contains an interval of 2 months. In addition, Lechner (2009) employs another estimator for this application: the inverse probability weighting (IPW) estimator. The results of these studies show that participating in a training course or receiving a temporary wage subsidy for two periods compared with remaining unemployed for two periods has positive effects on unsubsidised employment (approximately 10 percentage points 20 months after the end of the two periods). Using a timing-of-events approach, Graversen (2004) analyses the effects of sequences of different ALMPs (private sector employment programmes, classroom)

¹⁸ Training courses consist of basic courses, language courses, computer courses, further vocational training and courses for specific occupations but not occupational retraining. Employment programmes may be offered by public or private institutions and they should be similar to regular employment but should be neutral in terms of competition. Temporary wage subsidies for employees must target regular, but temporary, employment (Gerfin/Lechner 2002).

training and other programmes)¹⁹ on the transition rate from welfare to regular employment for Danish welfare recipients for the period between 1994 and 1998. The results show that lock-in effects arise for all first and second programmes during the programme periods. In addition, treatment effects after the first programme are positive for private sector employment programmes (an approximately 340% increase in the transition rate), public sector employment programmes (an approximately 150%) increase in the transition rate) and classroom training (an approximately 80% increase in the transition rate). However, the treatment effects after the second programme are mostly negative. Using the IPW estimator, Lechner/Wiehler (2013) evaluate sequences of five different states for unemployed individuals in Austria for the period between 2000 and 2002: unemployment (UE), orientation measure (OM), qualification measure (QM), active job search (AJS) and course subsidies (CS).²⁰ The authors consider both the timing and the order of the programmes. They find that earlier programme allocation has negative effects on unemployment (approximately -6 to -10 percentage points) for all programmes and target populations 4 years after initial entry into unemployment. However, the effects of the order of the programmes suggest that an active job search is more beneficial after a qualification measure: negative effects on unemployment (approximately -8 percentage points) arise 4 years after initial entry into unemployment.

5 Implementation and Method

Because the definition of periods is essential for analyses of sequences, I first describe the definition of states and sequences for the dynamic window approach of Dengler (2013). Second, I describe the dynamic matching approach and its assumptions.

5.1 Implementation of Dynamic Window Approach

I consider classroom training (denoted CT in the analysis) and all possible combinations of classroom training and UB-II-receipt (denoted UBII in the analysis). Moreover, I also consider One-Euro-Job participation (denoted 1EJ in the analysis). Final-

¹⁹ Private sector employment programmes consist of ordinary job training and individual job training to provide individuals with the experience of regular employment. The employer receives a wage subsidy in ordinary job training, whereas the earnings in individual job training consist of welfare benefits and an employment supplement. Public sector employment programmes include ordinary job training, individual job training and employment projects. Public job training programmes occur in the public sector but share similar characteristics with private sector employment programmes. However, employment projects are created by the municipality, consist of work that would not be performed otherwise (e.g., nature preservation) and only pay the amount of the welfare benefit (Graversen 2004).

²⁰ Orientation measures assess one's individual situation and aptitude. Qualification measures and active job search are components of the training programmes. Active job search should improve participants' job acquisition skills such as interview training. Qualification measures are offered by the Public Employment Service and they consist of both basic skill courses and training with a vocational degree. By contrast, course subsidies include financial support for courses that are offered by external providers (Lechner/Wiehler 2013).

ly, I analyse the following sequences: (CT,CT) versus (UBII,UBII), (CT,CT) versus (UBII,CT), (CT,UBII) versus (UBII,CT) and (CT,1EJ) versus (UBII,1EJ).

In this paper, I consider three periods (t=0, 1, 2) and three different states (UBII, CT, 1EJ). Figure 2 presents the possible states and sequences. In period 0, all individuals have the same state: UB-II-receipt without contributory employment (entry into the sample). In period 1, an individual can participate in classroom training or merely receive UB II without participating in an ALMP. Again, in period 2, an individual can participate in classroom training (or in a One-Euro-Job) or merely receive UB II without participating in an ALMP. Thus, I obtain six different sequences: (CT,CT), (UBII,UBII), (CT,UBII), (UBII,CT), (CT,1EJ), and (UBII,1EJ).

In contrast to the studies (co-)authored by Lechner (e.g., Lechner/Miquel (2010)), who defines a period as an interval of time, I use the dynamic window approach of Dengler (2013), which consists of a first start and individual window and a second start and individual window (Figure 3). Following the timing of events approach of Sianesi (2004), I use a start window of up to 122 days, in which an individual can begin the considered programme (CT or 1EJ) or not (UBII). The individual window takes 30 days (or 183 days)²¹ because of the average duration calculated by the different durations of the different types of classroom training set by law.²²

I consider classroom training versus UB-II-receipt as the first state: individuals can begin their first valid classroom training up to 122 days after entry into the sample or can receive UB II without beginning a valid programme up to 122 days after entry into the sample (first start window). Programmes are valid if they occur during the same welfare spell, i.e., permanent UB-II-receipt (with gaps of less than 31 days) without contributory employment (exit condition). If the individual gets the state UBII, I calculate random programme starts as random durations of time after the entry date that are randomly drawn from the empirical distribution of durations of classroom training. To determine the end of the first period and the beginning of the second period, I construct a first individual window by adding 30 days to the (random) starts of the first programmes. I create the first individual window to guarantee comparability and to decrease variation across the sequences with respect to duration.

For the second state, I consider three different states: CT, UBII and 1EJ. Individuals can begin valid classroom training (or a valid One-Euro-Job) up to 122 days after the end of the first individual window or can receive UB II without beginning a valid programme up to 122 days after the end of the first individual window (second start

²¹ Because I also consider One-Euro-Jobs for the second state only, I use 183 days as the individual window for the second individual window because of the average planned duration of One-Euro-Jobs (Dengler 2013).

²² By law, application training lasts up to 2 weeks, aptitude test up to 4 weeks and skill training up to 8 weeks. This yields an average duration of 32.6 days which I round down to 30 days (1 month). The average duration of classroom training in the sample also corresponds to these 30 days.

window). Again, if the individual achieves the state of UBII, I calculate random programme starts from the random durations of programme starts of classroom training or One-Euro-Jobs during the second period. Furthermore, I add 30 days (in the case of classroom training) or 183 days (in the case of One-Euro-Jobs) to the (random) starts of the second programmes to obtain the second individual window. As I measure outcomes since the (random) starts of the second programmes, the second individual window exists only for the sake of completeness.

If the first state consists of a valid first programme (CT), the second state consists of a valid programme (CT or 1EJ) only if it is the second programme after entry into the sample and if it starts during the second start window. Thus, the sequences (CT,CT) and (CT,1EJ) consist of the first programme after entry into the sample in the first period and the second programme after entry into the sample in the second period. If the first state is UBII, the second state consists of a valid programme only if it is the first programme after entry into the sample and if it starts during the second state window. Thus, the second state consists of a valid programme only if it is the first programme after entry into the sample and if it starts during the second start window. Thus, the sequences (UBII,CT) and (UBII,1EJ) consist of the first programme in the second period. Consequently, no programmes are allowed in the first individual window for the sequences (CT,CT), (CT,1EJ), (UBII,CT), and (UBII,1EJ); thus, I also exclude programmes for the sequences (UBII,UBII) and (CT,UBII) from the first individual window.²³

5.2 Method

I use the dynamic matching approach of Robins (1986), Lechner (2004), Lechner (2008), Lechner (2009) and Lechner/Miquel (2010) to analyse the effects of sequences. This approach solves dynamic selection problems during a sequence by considering intermediate variables that are influenced by the first state and that influence the second state.²⁴

In this paper, I consider three periods and three different states (see section 5.1). However, to simplify the notations used in the method section, I consider three periods (0, 1, 2) and only two different states (UBII and CT). All individuals are in the same state in the first period 0: UB-II-receipt without contributory employment. The vector of random variables $S = (S_0, S_1, S_2)$ describes the sequence in which an individual participates up to period 2, measured at the start of each period. A particular realisation of S_t is denoted by $s_t \in \{0,1\}$. A bar below a variable such as $\underline{s}_2 = (s_1, s_2)$ denotes the history of variables up to period 2. In period 1, an individual can take part in CT or UBII. Again, in the second period, an individual can take part in CT or UBII. Thus, I get four different sequences for this example: (CT,CT), (CT,UBII), (UBII,CT) and (UBII,UBII).

²³ First programmes occurred between October 2005 and January 2007 and second programmes occurred between November 2005 and June 2007.

²⁴ See also Dengler (2013) for a detailed overview of the dynamic matching approach.

Similar to the static approach, I estimate the average causal effects of sequence k $(\underline{s}_{\tau}^{k})$ versus sequence h $(\underline{s}_{\tau}^{h})$ up to period τ (=2) for a specific subpopulation j $(\underline{s}_{\tau}^{j})$ in period $\tilde{\tau}$ (=1) by calculating the differences between the potential outcomes of sequence k and sequence h measured at the start of period t (=2), $Y_{t}^{\underline{s}_{2}^{h}}$ and $Y_{t}^{\underline{s}_{2}^{h}}$.

$$\begin{aligned} \theta_t^{\underline{S_r^k},\underline{S_\tau^h}}(\underline{s_{\tilde{\tau}}^j}) &\coloneqq E\left(Y_t^{\underline{S_\tau^k}} \middle| \underline{S_{\tilde{\tau}}} = \underline{s_{\tilde{\tau}}^j}\right) - E(Y_t^{\underline{S_\tau^h}} \middle| \underline{S_{\tilde{\tau}}} = \underline{s_{\tilde{\tau}}^j}) \\ 0 &\leq \tilde{\tau} \leq 2, \ 1 \leq \tau \leq 2, \ \tilde{\tau} \leq \tau, \ k \neq h, \ k,h \in (1, \dots 2^{\tau}), \ j \in (1, \dots 2^{\tilde{\tau}}) \end{aligned}$$

The most interesting effects are the dynamic average treatment effects on the treated (DATET): $\theta_2^{S_2^k,S_2^h}(s_1^k)$. Thus, I compare sequence k up to period 2 with the sequence h up to period 2 for individuals participating in the first state of sequence k in period 1. For example, the DATET of sequence (CT,CT) versus sequence (UBII,UBII) for individuals participating in classroom training in the first period. I use a sequential version of the propensity score matching estimator for the estimation.²⁵ The sequential matching aims to match sequence k (e.g., (CT,CT)) and sequence h (e.g., (UBII,UBII)) to the subpopulation j (e.g., the population participating in classroom training in the first period (CT)). Thus, for individuals participating in classroom training in the first period, I compare whether participating in sequence (CT,CT) or participating in sequence (UBII,UBII) is more beneficial.

To identify the effects, assumptions similar to the static approach must hold (Roy 1951; Rubin 1974). First, the Stable Unit Treatment Value Assumption (SUTVA) must hold. That is, an individuals' treatment and outcomes must not depend on other individuals' treatment and outcomes (Rubin 1980). This assumption is likely to hold if the considered programmes are not large scale (Frölich 2004). Second, the Weak Dynamic Conditional Independence Assumption (WDCIA) for dynamic propensity score matching (e.g., Lechner/Miquel (2010)), which is similar to the static Conditional Independence Assumption (CIA) of Lechner (1999), must hold. The CIA states that potential outcomes are independent of the treatment if all covariates that jointly influence the treatment and outcomes are controlled for (Caliendo/Kopeinig 2008). If the CIA is valid conditional on covariates, it is also valid conditional on balancing scores such as propensity scores (Rosenbaum/Rubin 1983). While the first part of the WDCIA comprises the usual CIA, its second part considers intermediate outcomes. First, potential outcomes are independent of the treatment in period 1 conditional on pre-treatment covariates (covariates of period 0 (X₀)). Second, potential outcomes are independent of the treatment in period 2 conditional on the treatment in period 1, the covariates of period 0 (X_0) and the covariates and outcomes of period 1 (X_1). Furthermore, the WDCIA of Lechner/Miquel (2010) includes the usual Common Support Requirement (CSR): individuals with the same characteristics (X_0

²⁵ For a short matching protocol, see Table A-1 in the appendix.

and X_1) must have a positive probability participating in all of the considered sequences and subpopulations.

6 Data and Descriptives

Because the dynamic matching approach is very data hungry, I use rich administrative data from the German Federal Employment Agency. First, I describe the data and discuss the identification of effects based on the WDCIA. Second, I provide some descriptive statistics.

6.1 Data and Identification

I use administrative data from the German Federal Employment Agency that have been prepared for scientific use by the Institute for Employment Research.²⁶ I draw a rich inflow sample of all UB-II-recipients without contributory employment for the period between October 2005 and September 2006.²⁷ In addition, the individuals are aged 18 to 57 years and had not received UB II 3 months before their entry date into the sample.^{28,29}

I need rich data on covariates and intermediate outcomes for the justification of the WDCIA. Thus, I have to observe all covariates and intermediate outcomes that jointly influence the treatment and potential outcomes. Assignment into programmes depends on legal requirements, selection by case workers and self-selection by welfare recipients (Lechner/Miquel/Wunsch 2011). First, legal requirements for assignment into ALMPs are an important factor. In the case of classroom training and One-Euro-Jobs, the participants must be in UB-II-receipt, which is the sample condition. Additionally, young welfare recipients who are less than 25 years of age are a specific target group in the UB-II-system. Thus, I include age as a covariate.

Second, the case workers assign unemployed welfare recipients into programmes based on a detailed profiling process (Jacobi/Kluve 2007). This assignment into programmes depends on the welfare recipient's employment prospects and success in completing a specific programme and regional labour market conditions (Lechner/Miquel/Wunsch 2011). Thus, I include socio-demographic variables (i.e., age, education and nationality), variables related to the labour market history (i.e., employment history, unemployment history and ALMP history) and variables related to the last contributory employment (i.e., time since last contributory job, status,

²⁶ I use data from the Integrated Employment Biographies (IEB) and the UB-II-Receipt History ('Leistungshistorik Grundsicherung' (LHG)).

²⁷ Data from local authorities ('zugelassene kommunale Träger') are not included because of data collection problems. In these 69 districts, the Federal Employment Agency did not administer UB II. For approximately 13% of unemployed welfare recipients between 2005 and 2008, UB II was administered by local authorities (Department for Statistics of the Federal Employment Agency 2014).

²⁸ For additional data preparations, see section 6.1 in Dengler (2013) for more details.

²⁹ The upper limit of the observation window is October 2009 and the upper limit of employment information is December 2008.

wage and industry type). Furthermore, I use information on household variables (i.e., partner, children, equivalent household income from welfare and variables capturing the partner's socio-demographic information and labour market history), because household context and partner employment prospects can also influence individuals' employment prospects and success in completing a specific ALMP. Furthermore, I consider regional labour market variables (i.e., unemployment rate, longterm unemployed per unemployed, vacancies per unemployed), because these variables also contribute to individuals' employment prospects (Lechner/Miquel/Wunsch 2011). All of these variables are measured before or at entry into the sample. In their case worker survey, Achatz et al. (2009) also report important assignment criteria for short-term training that are mostly captured by the variables included in this paper. However, the potential participants' motivation is an important assignment criterion for case workers that I measure only indirectly by labour market history.

Third, self-selection by welfare recipients into programmes could be also an important assignment factor. Normally, case workers decide together with the unemployed welfare recipient about participating in a programme and in which type of programme. The decision of a welfare recipient to participate in a programme or not is similar to that of the case workers, although additional reasons for participating or not could be important, e.g., the welfare recipient does not want to reduce his/her leisure time (Lechner/Miquel/Wunsch 2011). However, controlling for rich individuals' employment prospects and regional labour market variables will also capture welfare recipients' self-selection. Furthermore, case workers have the final decision about programme assignment (Yankova 2010) and if an unemployed welfare recipient refuses to participate in a programme, he/she also risks his/her benefit receipt due to sanctions. Thus, welfare recipients' self-selection into ALMPs is limited anyway.

Unobserved variables such as motivation, personality traits or information on the case workers are not controlled for, but are indirectly captured by rich information on individuals' (un-)employment histories. Caliendo/Mahlstedt/Mitnik (2014) also suggest that administrative data with detailed labour market history information are sufficient to control for usually unobserved variables.

To justify the WDCIA, I also need information on intermediate variables that are influenced by the first state and that influence the second state. Thus, I need information on time-varying variables that drive assignment to a second programme. First, Lechner/Wiehler (2013) assume that a case worker's decision to assign an unemployed welfare recipient to a further programme is based on intermediate employment prospects of the welfare recipients. Second, intermediate financial or individual and household criteria influence assignment to a second programme. The intermediate variables are measured before or at period 2.

Because the first individual window takes only 30 days, I only consider intermediate outcomes that may change during this time period (e.g., I do not consider education

as an intermediate outcome). Furthermore, because individuals must be in permanent UB-II-receipt (with gaps of less than 31 days) without contributory employment (exit condition), intermediate variables on employment prospects may change for the cumulated duration of UB-II-receipt to some extent and for the cumulated duration of minor employment measured 1 month³⁰ before period 2, but not for contributory employment. Equivalent household income from welfare in the month of period 2 captures not only intermediate financial aspects but also household aspects. Indeed, equivalent household income may change because of a change in household composition. Thus, I consider children at period 2 as part of the household or as individual aspects.³¹ Sanctions may be an additional important intermediate variable, but because of the incomplete data on sanctions before January 1, 2007, I cannot use this information. However, both sanctions and changes in household composition are indirectly captured by the equivalent household income. Thus, I can control for all of the observed intermediate outcomes that are influenced by the first state and that influence the second state.

I account for various outcomes that are measured 1 month after the (random) start of the programme in the second period. First, I include regular employment rate (unsubsidised contributory employment). Second, I consider ALMP outcomes on qualification programmes (short-term training and further vocational training) and One-Euro-Jobs as indicators for stepwise integration through additional programmes or programme careers. Third, I regard the avoidance of UB-II-receipt as an outcome. Indeed, individuals may leave UB-II-receipt if their earnings or the earnings of the household are sufficiently high (e.g., by taking up a regular employment) or if they drop out of the labour market.

6.2 Descriptives

Table 2 presents the sample sizes and some descriptive statistics for the sample and the considered subpopulations and sequences. More than 350,000 West German men and women are included in the sample. However, the number of observations for the considered sequences sharply decreases, especially for sequences consisting of two consecutive programmes (e.g., the observations for women in the sequence (CT,1EJ) decrease to 380). Nevertheless, the number of observations is sufficiently high for estimation.

The sample consists of equal proportions of men and women, but men participate more frequently in classroom training in the first period. Women also participate in the considered sequences to a lesser extent than men, especially in sequences with One-Euro-Jobs ((CT,1EJ) and (UBII,1EJ)). Turning to some descriptive results, I find that the average age ranges from 30 to approximately 34 years, except for the se-

³⁰ I use 1 month because my individual window is 30 days. Otherwise, the intermediate outcomes would extend into the pre-treatment period (before period 1).

³¹ The variable pregnant can be constructed at period 2. However, the number of observations is too small for estimation.

quence (CT,1EJ) for men. In general, women are more likely than men to have children: for the considered sequences, the share of children is nearly twice as high for women as for men. Moreover, the average cumulated duration of minor employment 5 years before entry is higher for women than for men (men spent more than 100 days and women spent more than 200 days in minor employment), but for the sequences consisting of One-Euro-Jobs, the average durations are slightly lower for women. The average cumulated duration of UB-II-receipt 1 year before entry is approximately 1 month for men and approximately 23 to 26 days for women. However, the average duration of UB-II-receipt is higher for the sequences (CT,1EJ) and (UBII,1EJ). In addition, women more frequently had no contributory employment before entry (approximately 30 to 40%) compared with men (approximately 20 to 30%), and the average equivalent household income from welfare³² in the month at entry is more than 380 Euros for men and approximately 300 to 400 Euros for women.

The intermediate variables present the average cumulated duration in minor employment 1 month before period 2 and a dummy for being in minor employment at period 2. The average duration in minor employment is higher for women than for men (approximately 2 to 5 days) and they are more frequently in minor employment at period 2. The average cumulated duration of UB-II-receipt 1 month before period 2 is approximately 1 month for both men and women. This result is not surprising because the programmes must be in the same welfare spell, i.e., permanent UB-IIreceipt (with gaps of less than 31 days) without contributory employment (exit condition). The intermediate variable on children at period 2 shows a slightly higher share of children at period 2 than at entry for nearly all subpopulations and sequences, especially among women in (UBII, UBII). The intermediate variable on equivalent household income from welfare in the month before period 2 is approximately 100 to 190 Euros higher than the equivalent household income from welfare in the month at entry. One possible reason for this result may be that welfare recipients leave their minor employment and thus receive more income from welfare or that welfare recipients still have some income from other sources or employment at entry into UB II.

Table 3 presents the outcomes for all controls, treated individuals, matched controls and matched treated individuals 12 months after the programme start in the second period for the considered sequences. First, I find some positive selection for the sequence (CT,CT) because a higher proportion of the matched controls of the sequence (UBII,UBII) are in regular employment than all controls of the sequence (UBII,UBII). Second, a higher proportion of participants in the sequences (CT,CT) and (CT,1EJ) are in ALMPs, such as short-term training and One-Euro-Jobs, com-

³² To calculate the equivalent household income from welfare, I use the OECD modified scale, which assigns the head of household a weight of 1, each additional adult a weight of 0.5 and each child a weight of 0.3.

pared with participants in their respective counterparts ((UBII,UBII), (UBII,CT), and (UBII,1EJ)).

7 Results

In this section, I present the four DATET for different labour market outcomes. I also consider heterogeneous effects by calculating the DATET only for older individuals aged at least 25 years. Furthermore, I perform a robustness check by applying a different matching algorithm: caliper matching.

7.1 Overall Results

Before I discuss the four DATET for West German men and women, I present the matching quality by calculating the mean standardised absolute bias (MSB).

7.1.1 Matching Quality

For each subgroup (i.e., West German men and women), I check the matching quality after each of the three matching steps (dynamic matching 1 to 3) and the final matching quality of the considered sequences at the end (final matching) by calculating the MSB.³³ The MSB is defined as the distance in the marginal distribution of the covariates for all of the covariates that are included in each of the probit models. The MSB is reduced if the matching is successful. Caliendo/Kopeinig (2008) suggest that a reduction of the bias to 3 to 5 percent after matching is sufficient.

For each of the three matching steps, I calculate the MSB for each covariate included in each of the probit models before and after matching. As I match the sequence k to the sequence h via these three matching steps, I also calculate the MSB for the final matching quality of the sequence k to the sequence h before and after matching by including all of the covariates.³⁴ Because the CSR also must hold, some observations are dropped before each matching step.

Table 4 includes the MSB before and after matching for West German men and women for each of the three matching steps. Before matching, the MSB ranges from approximately 4 percent to approximately 16 percent for dynamic matching steps 1 to 3 for West German men and women. After matching, the MSB is mostly below 3 percent for dynamic matching steps 1 to 3.

³³ I also calculate the means for each covariate that is included in the probit models between the treated individuals and (matched) controls before and after matching and the pvalues of the t-test of the differences between the means. Thus, I can check the matching quality after each of the three matching steps and the final matching quality of the considered sequences at the end for each subgroup and covariate. Due to space restrictions, results are only available upon request.

³⁴ Not all of the covariates must be included in all of the probit models of the three matching steps. Thus, this may lead to poorer final matching quality, as I consider all MSBs for all the covariates.

Table 4 also provides the MSB before and after matching for the final matching of the considered sequences: The MSB for the final matching is below 5 percent for (CT,UBII) versus (UBII,CT) and for (CT,CT) versus (UBII,CT) (for West German women, the MSB is slightly above 5%, at 5.2%). The MSB for (CT,CT) versus (UBII,UBII) and for (CT,1EJ) versus (UBII,1EJ) also considerably reduced after matching, reaching approximately 6 percent, but the MSB after matching for the final matching remains very high for West German women for (CT,CT) versus (UBII,UBII), at approximately 11.4 percent.

Previous studies on sequences that use the dynamic matching approach do not consider the matching quality for the three matching steps and for the final matching; thus, no recommendations for a successful final matching are available. Because the sequences are not directly matched but are matched only via three dynamic matching steps to the subpopulation of classroom training, the MSB after matching for the sequences may not be reduced below the level of 3 percent to 5 percent. The MSB for the final matching is only an approximation of the final matching quality. Thus, future research on the dynamic matching approach must quantify which reductions may be considered sufficient for the final matching.

However, I suggest that a reduction below 6 to 10 percent for the MSB after matching of the final matching may be sufficient. Thus, in summary, the matching quality for each of the three matching steps and the final matching is very good, but not for West German women for (CT,CT) versus (UBII,UBII).

7.1.2 Overall Effects

In this section, I present the DATET for the considered sequences for West German men and women: $\theta_2^{(CT,CT)(UBII,UBII)}(CT)$, $\theta_2^{(CT,UBII)(UBII,CT)}(CT)$, $\theta_2^{(CT,CT)(UBII,CT)}(CT)$ and $\theta_2^{(CT,1E])(UBII,1E]}(CT)$. I estimate the DATET for the following outcomes which are measured 1 month after the (random) start of the programme in the second period: regular employment rate, short-term training, further vocational training, One-Euro-Jobs and no UB-II-receipt. All of the outcomes are available for up to 26 months after the start of the programme in the second period, but regular employment rate is only available up to 18 months after the start of the programme in the second period.

7.1.2.1 DATET for (CT,CT) versus (UBII,UBII)

First, I present the basic comparison, the DATET for (CT,CT) versus (UBII,UBII), for West German men and women participating in classroom training in the first period in Figure 4 through 8 for the different outcomes.

Figure 4 presents the DATET on regular employment rate. Initially negative but not significant regular employment effects emerge for West German men participating in classroom training in the first period if they participate in the sequence (CT,CT) versus the sequence (UBII,UBII). The effects turn positive in the medium run (approxi-

mately 7 percentage points), but they are mostly not well-determined. However, well-determined regular employment effects arise for West German women 8 to 18 months after the (random) start of the programme in the second period-up to approximately 13 percentage points. One reason for the substantial effectiveness of the sequence (CT,CT) for West German women may be that West German women may be easily activated after the Hartz IV reform if their partners lost their jobs. Before the reform, social assistance recipients and members of unemployment assistance households were often not activated.³⁵ Bergemann/Van den Berg (2008) also find in their survey on the effects of ALMPs for European countries that the effects of skill training are higher for women compared with men, especially in countries with low female labour force participation. In addition, the authors show that the gap in labour force participation between men and women in West Germany remains very high (18 percentage points in 2004). The authors suggest that a possible reason for the substantial effectiveness of ALMPs for women may be that the female labour supply is more elastic than the male labour supply because women have more outside options (e.g., child bearing and/or child caring) than men.

In summary, two classroom trainings compared with two periods of UB-II-receipt are quite effective in terms of regular employment for West German women participating in classroom training in the first period. Thus, two classroom trainings may increase human capital, enhance one's job search activities and send positive signals to employers. Moreover, lock-in effects do not play a major role in classroom training because classroom training is short in duration and because it does not reduce the time available for job search. However, the effects for West German women may be biased because the final matching quality was not very good. I assume that the results on regular employment rate are upwardly biased because I find positive selection for the sequence (CT,CT): a higher proportion of the matched controls of the sequence (UBII,UBII) are in regular employment compared with all of the controls of the sequence (UBII,UBII).

In qualitative terms, my results are similar to those presented in the existing literature on classroom training that evaluates participating in a single classroom training compared with non-participation. For example, Wolff/Jozwiak (2007) find no initial lock-in effects and positive employment effects of classroom training (approximately 3 to 4 percentage points) that are smaller than the employment effects I find.

The effects on the avoidance of UB-II-receipt are mostly not well-determined (Figure 5). Thus, I cannot conclude that individuals participating in classroom training in the first period leave welfare if they participate in the sequence (CT,CT) versus the sequence (UBII,UBII). The positive regular employment effects for women may indi-

³⁵ See Dengler (2013).

cate that women may receive low wages that are not sufficient for their households to leave welfare.³⁶

Another reason for these results may be that the participants in the sequence (UBII,UBII) leave UB-II-receipt after the two periods for different reasons, but not because they obtain regular employment. Thus, I do not observe significant differences for no UB-II-receipt between participants of the sequence (CT,CT) and participants of the sequence (UBII,UBII). I also consider the DATET on the outcome without any status.³⁷ Not only individuals who are unavailable for employment or activation according to Article 10 SC II³⁸ but also individuals who are ill for more than 42 days or who are in retraining may be considered without any status. The effects on the outcome without any status are well-determined and negative (approximately -5 to -13 percentage points) for females participating in classroom training in the first period if they participate in the sequence (CT,CT) versus the sequence (UBII,UBII). Thus, positive effects on the outcome without any status emerge for female participants with the sequence (UBII,UBII). For example, women may enter UB-II-receipt because their partners have lost their jobs, but women may not be capable of working because of a pregnancy or childcare duties. Thus, these women obtain the status without any status and leave UB-II-receipt and the labour market after some time. Lechner/Whieler (2011) also find that Austrian ALMPs are highly effective for women. Controlling for pregnancies, they conclude that programme participants postpone pregnancies, whereas non-participants engage in childbearing, leave the labour force and have lower employment rates. Thus, the observed effectiveness of ALMPs for women stems from the greater number of outside options for women (such as pregnancies) compared with men.

Because programme careers can arise but further programmes might be necessary for a successful stepwise integration into regular employment, I also estimate the DATET for (CT,CT) versus (UBII,UBII) on certain ALMP outcomes: One-Euro-Jobs, short-term training and further vocational training³⁹. Consecutive One-Euro-Jobs may reflect a programme career, as One-Euro-Jobs are a last resort: individuals

³⁶ I also estimate the DATET on part-time regular employment rate (where an individual is part-time employed if his/her working hours are less than the common working hours in collective bargaining agreements or firms), but I mostly do not find well-determined effects. Furthermore, the data quality for part-time and full-time employment is not very good. Because of a change in the reporting process of contributory employment in 2011, the share of full-time employees is likely overestimated for past periods and, therefore, is likely overestimated in the considered data.

³⁷ The results are available upon request.

³⁸ In general, UB-II-recipients must be available for every job unless they must care for children under 3 years of age or other family members, unless they cannot perform a specific job owing to physical, mental, or psychological reasons, or unless the job would impede the performance of a former job that includes specific physical efforts.

³⁹ Further vocational training includes not only short qualification programmes (up to 1 year) but also long retraining programmes (up to 3 years) that result in a vocational training degree (Bernhard/Kruppe 2012).

may not find regular employment after the sequence of two classroom trainings and they may be assigned to One-Euro-Jobs, as the only alternative for being 'employed'. Consecutive short-term training may reflect both programme career and stepwise integration. However, consecutive further vocational training may indicate stepwise integration because it provides specific professional skills or even a vocational training degree for welfare recipients.

Figure 6 presents the DATET on One-Euro-Jobs. The effects on One-Euro-Jobs for West German men are positive and significant (approximately 4 to 5 percentage points) from 6 to 11 months after the (random) start of the programme in the second period. For West German women, I also find positive and well-determined effects (approximately 2 to 4 percentage points), but only 3 to 6 months after the (random) start of the programme in the second period. These results indicate that individuals participating in classroom training in the first period participate in One-Euro-Jobs in the first year after the (random) start of the programme in the second period if they participate in the sequence (CT,CT) versus the sequence (UBII,UBII). The results may reflect programme careers.

Figure 7 shows the DATET on short-term training. I find some well-determined and positive effects on short-term training in the first months after the (random) start of the programme in the second period for men and women participating in classroom training in the first period if they participate in two classroom trainings versus two periods of UB-II-receipt. At the beginning, the effects on short-term training are substantial (up to approximately 70 percentage points in the first month) because the participants remain in the second classroom training for the first 3 months. However, the effects decrease to approximately 4 percentage points after 4 months and they are mostly not well-determined afterwards. Thus, the results do not indicate any stepwise integration or programme careers consisting of short-term training.

Figure 8 provides the DATET on further vocational training. In the medium run, positive and significant effects on further vocational training (up to approximately 4 percentage points) arise for West German men and women participating in classroom training in the first period if they participate in the sequence (CT,CT) versus the sequence (UBII,UBII). Thus, the results indicate stepwise integration.

In summary, positive regular employment effects arise for individuals participating in classroom training in the first period if they participate in two classroom trainings versus two periods of UB-II-receipt, especially for West German women. However, I do not find positive and well-determined effects on the avoidance of UB-II-receipt. The effects on One-Euro-Jobs and further vocational training indicate both stepwise integration and programme careers.
7.1.2.2 DATET for (CT,UBII) versus (UBII,CT)

Figure 9 through 13 present the DATET for the sequence (CT,UBII) versus the sequence (UBII,CT) (i.e., the effects of timing) for West German males and females participating in classroom training in the first period.

The effects on regular employment rate are only well-determined and positive at the very beginning (Figure 9): I find positive regular employment effects (approximately 1 to 3 percentage points after 1 to 3 months) for West German males and West German females participating in classroom training in the first period if they do not wait for to receive classroom training in the second period. Thus, participating in classroom training to participate in classroom training in a second period, but after the first 3 months, no well-determined effects emerge.

The results are in contrast to Dengler (2013) who analyses the effects of timing for One-Euro-Jobs (i.e., (1EJ,UBII) versus (UBII,1EJ)) for participants in One-Euro-Jobs in the first period and finds that West German women are better off if they wait for their first One-Euro-Job. The result may be observed because West German women may be able to obtain a more appropriate One-Euro-Job that is in line with their former skills and tasks if they wait. Because classroom training may be more quickly available and offered more frequently in shorter time intervals – in contrast to One-Euro-Jobs – individuals may not need to wait to receive their first classroom training to obtain appropriate classroom training.

Figure 10 presents the DATET for (CT,UBII) versus (UBII,CT) on avoiding UB-II-receipt: the effects are mostly well-determined and positive, especially in the first year after the (random) start of the programme in the second period for West German men and women (approximately 2 to 5 percentage points). For men, positive and significant effects are also observed after 21 to 26 months. One reason that individuals may leave welfare receipt is to begin regular, gainful employment, but I only observe positive effects on regular employment rate in the first 3 months for West German men and for West German women. Thus, other reasons may explain the positive effects on avoiding UB-II-receipt, such as leaving the labour market or leaving UB-II-receipt because of the work test function of classroom training or receiving income from other sources (e.g., wages from a partner that are sufficiently high to avoid UB-II-receipt).⁴⁰

Figure 11 shows the DATET on One-Euro-Jobs. The effects on One-Euro-Jobs are mostly not well-determined for West German women. However, negative and significant effects arise for West German men (approximately 1 to 2 percentage points). Thus, I can conclude that males participating in classroom training in the first period

⁴⁰ I also estimate the effects on subsidised employment, but I find mostly no welldetermined effects or even negative effects.

are not subject to programme careers if they participate in the sequence (CT,UBII) versus the sequence (UBII,CT).

Figure 12 provides the DATET on short-term training: negative, well-determined effects emerge in the first 3 months because the participants in the sequence (UBII,CT) are still in their second period of classroom training, whereas the participants in the sequence (CT,UBII) are not. However, the effects become mostly not significant and close to zero afterwards.

Figure 13 presents the DATET on further vocational training: I find negative and well-determined effects for West German men for up to 19 months after the (random) start of the programme in the second period. However, only sporadically significant and negative effects emerge for West German women. Thus, males participating in classroom training in the first period avoid stepwise integration through further vocational training if they participate in the sequence (CT,UBII) versus the sequence (UBII,CT).

In summary, I find some positive regular employment effects in the beginning; however, afterwards, there are no beneficial effects of starting classroom training immediately versus waiting for classroom training in a second period. Thus, starting classroom training immediately may lead to faster integration into employment at the beginning because of reduced matching efforts or human capital losses. Because the effects of leaving UB-II-receipt are well-determined and positive, factors other than obtaining regular employment may be responsible for these effects. For instance, because classroom training tests welfare recipients' willingness to work, they may leave the labour market. Males participating in classroom training in the first period also avoid programme careers or stepwise integration if they immediately participate in classroom training versus if they participate in classroom training in the second period.

7.1.2.3 DATET for (CT,CT) versus (UBII,CT)

Figure 14 through 18 present the effects of participation in multiple programmes, i.e., the sequence of two classroom trainings versus the sequence of UB-II-receipt followed by classroom training. However, I do not find well-determined effects, or I find such effects only sporadically. No well-determined effects on regular employment rate for West German men and women emerge: only one significant and positive effect (approximately 3 percentage points after 2 months) arises for West German women (Figure 14). Figure 15 presents the DATET on avoiding UB-II-receipt, which are also not well-determined: only males participating in classroom training in the first period show some sporadic positive effects on leaving UB II in the medium run (approximately 7 percentage points). The effects on ALMPs (One-Euro-Jobs, short-term training and further vocational training) are also mostly not well-determined (Figure 16 through 18).

7.1.2.4 DATET for (CT,1EJ) versus (UBII,1EJ)

Figure 19 through 23 show the DATET for the sequence (CT,1EJ) versus the sequence (UBII,1EJ) for individuals participating in classroom training in the first period. Again, these effects reflect the effect of participation in multiple programmes because participation in two programmes (i.e., classroom training followed by a One-Euro-Job) is compared with participation in only one programme (i.e., a One-Euro-Job).

Figure 19 presents the effects on regular employment rate. Most of the effects are not well-determined. The effects of leaving UB-II-receipt are also mostly not well-determined; however, for West German men, some positive effects (approximately 6 to 8 percentage points) emerge (Figure 20).

Figure 21 through 23 present the effects on One-Euro-Jobs, short-term training and further vocational training; however, only sporadically significant effects are observed. Specifically, I find positive but not well-determined effects on further vocational training (significant effects emerge only after 9 and 24 months, at approximately 4 percentage points) for West German women.

In summary, most effects for the sequence (CT,1EJ) versus the sequence (UBII,1EJ) are not well-determined. Thus, the potential positive effects of the strategic mix of first classroom training followed by a One-Euro-Job and the positive effects of waiting for a more appropriate One-Euro-Job in the second period seem to outweigh each other.

7.2 Heterogeneous Effects

In this section, I only consider individuals aged at least 25 years. Because the treatment effects may differ between different subgroups such as age groups, the heterogeneous effects may differ from the effects for the entire sample.⁴¹

Young adults aged 15 to 24 years are a special target group of the SC II that have to be placed without delay into employment, vocational training, or, as a last resort, a One-Euro-Job.⁴² Furthermore, young adults may not be registered as unemployed for more than 3 months, leading to a potential widespread assignment of young adults into classroom training because such programmes are cheap and short (Wolff/Jozwiak 2007). Thus, the resulting poor matching of young adults to classroom training may diminish the effectiveness of the classroom training. Indeed, pre-

⁴¹ As the number of observations are too small for young adults aged 18-24 years, I estimate the DATET for individuals aged at least 25 years and compare the results to DATET of the entire sample of all individuals (chapter 7.1.2).

⁴² Since April 2012, job centres are no longer required to place young adults in One-Euro-Jobs without delay.

vious studies on young adults mostly find no beneficial effects of classroom training. $^{\rm 43}$

Table 5 presents the MSB before and after matching for West German men and women aged at least 25 years. In general, the results for matching quality are very similar to the results presented in section 7.1.1 The MSB after matching for the final matching is generally below 8 percent, but not for West German women for (CT,CT) versus (UBII,UBII) and for (CT,1EJ) versus (UBII,1EJ).

Figure 24 through 31 present the DATET on regular employment rate for all individuals of the entire sample and for older individuals aged at least 25 years, with 95 percent confidence bands plotted separately for West German men and women. Thus, no significant differences exist if the confidence bands for the entire sample overlap with the confidence bands for the older-aged individuals.

For older West German men, the effects on the regular employment rate are more significant and higher (approximately 6 to 10 percentage points) for the sequence (CT,CT) versus the sequence (UBII,UBII) than for all West German men (Figure 24). Thus, the sequence (CT,CT) may be less effective for young men in terms of regular employment. However, the differences are not significant. Figure 25 shows the effects on the regular employment rate for West German women for the sequence (CT,CT) versus the sequence (UBII,UBII), which are slightly smaller (approximately 6 to 10 percentage points) for older women than for all women. Thus, older-aged women may have slightly fewer outside options (e.g., child bearing or child caring), but the differences are not significant. In summary, the effects on not only regular employment rate but also all other outcomes are quite similar for all and older-aged individuals.⁴⁴ Furthermore, the differences in the effects between all individuals and older-aged individuals are not significant.

Figure 26 shows the effects of timing (i.e., the sequence (CT,UBII) versus the sequence (UBII,CT)) for all West German men and older-aged West German men on the regular employment rate. The effects are mostly not well-determined. Further, the initial positive and well-determined regular employment effects for all West German men are not observed for older-aged West German men. However, I find that the sequence (CT,UBII) versus the sequence (UBII,CT) may test individuals' willingness to work, leading to well-determined and positive effects on leaving UB-II-receipt for all West German men. For older individuals, the effects of leaving UB II are mostly smaller and less significant.⁴⁵ Thus, young adults seem more subject to the work test function of the sequence (CT,UBII) compared with the sequence

⁴³ See section 4.1 for an overview of studies on young adults and training programmes.

⁴⁴ Because of space restrictions, I do not present the figures on all other outcomes (no UB-II-receipt, One-Euro-Jobs, short-term training and further vocational training), but they are available upon request.

⁴⁵ Results are available upon request.

(UBII,CT). However, the differences between all West German men and older-aged West German men are not significant. Similar patterns of results are observed for West German women (Figure 27).

The regular employment effects of the sequence (CT,CT) versus the sequence (UBII,CT) are presented in Figure 28 for West German men and in Figure 29 for West German women, whereas Figure 30 and 31 show the regular employment effects for the sequence (CT,1EJ) versus the sequence (UBII,1EJ) for West German men and women, respectively. The effects are mostly not well-determined for older-aged individuals and the differences between all individuals and older-aged individuals are generally not significant.

In summary, the effects for individuals aged at least 25 years are quite similar to the effects for the entire sample. However, some differences are observed, but they are not significant. Thus, poor matching of young adults to sequences of classroom training does not take place.

7.3 Robustness Check

In the following section, I present the results of a robustness check in which I apply a different matching algorithm, namely, caliper matching, to determine whether my results are robust to the use of different matching procedures.

I apply caliper matching instead of using nearest-neighbour matching (one-to-one) with replacement in matching steps 1 and 3. For the mahalanobis matching in matching step 2, I also employ a caliper. Thus, in all three matching steps, I calculate the calipers by estimating the 95 percent percentile of the differences between the propensity scores of the treated individuals and those of the controls by using nearest-neighbour matching (one-to-one) with replacement. Thus, I drop the poorest 5 percent of the matches.

First, I compare the matching quality results in this robustness check with the matching quality results from section 7.1.1. Table 6 presents the MSB before and after matching for West German men and women based on caliper matching. The MSB before matching with caliper matching is very similar to the MSB before matching in section 7.1.1 However, the MSB after matching is generally increased for the dynamic matching steps 1 to 3, but the MSB is only slightly increased compared with the MSB after matching in section 7.1.1. Thus, the MSB after matching for the final matching is also slightly higher than its counterpart in section 7.1.1. In summary, caliper matching (95 percent percentile) does not perform better than nearestneighbour matching (one-to-one) with replacement and mahalanobis matching without any caliper. Rather, the matching quality is quite similar.

The results of the robustness check are presented in Figure 32 through 35 for the considered sequences and only for the outcome of regular employment rate. The

results on not only regular employment rate but also the other outcomes⁴⁶ are guite similar to the results presented in section 7.1.2. However, regarding the sequence (CT,CT) versus the sequence (UBII,CT), some positive and well-determined effects (approximately 6 percentage points) emerge for men approximately 1 year after the start of the programme in the second period, as well as some positive and welldetermined effects on leaving UB-II-receipt. In summary, the results from this robustness check are guite similar to the results presented in section 7.1.2.

Summary and Conclusions 8

Sequences of ALMPs are very common among welfare recipients in Germany, but most studies evaluating ALMPs do not analyse such sequences. Given that One-Euro-Jobs and classroom training are very common ALMPs and are very common components of sequences of ALMPs, Dengler (2013) analyses sequences of One-Euro-Jobs for men and women in West and East Germany and this paper analyses sequences of classroom training for West German men and women.

The paper analyses the effects of different sequences of ALMPs for individuals participating in classroom training in the first period. First, I consider a basic comparison: the sequence of two classroom trainings compared with the sequence of two periods of UB-II-receipt. Second, I analyse the effect of timing by comparing the sequence of initial classroom training followed by UB-II-receipt with the sequence of UB-II-receipt followed by classroom training. Third, I consider the effect of participation in multiple programmes by comparing participation in two programmes with comparing participation in only one programme (i.e., the sequence of two classroom trainings versus the sequence of UB-II-receipt followed by classroom training). Fourth, I analyse the effect of participation in multiple programmes for one additional, very common sequence: the sequence of classroom training followed by a One-Euro-Job compared with the sequence of UB-II-receipt followed by a One-Euro-Job. By using rich administrative data and a dynamic matching approach, I control for dynamic selection problems that may occur during a sequence. I implement the dynamic matching approach introduced by Lechner (2004), Lechner (2008), Lechner (2009) and Lechner/Miguel (2010), but I use a different definition of period based on the dynamic window approach. Furthermore, I control for the final matching quality of the considered sequences, which are matched via three single matching steps to the subpopulation.

In summary, participating in two classroom trainings compared with receiving UB II for two periods is highly effective in terms of regular employment outcomes, especially for West German women. In some cases, participation in multiple programmes should be avoided: participation in two classroom trainings has mostly no beneficial

⁴⁶ Because of space restrictions, I do not present the results for all outcomes but present the results for regular employment rate only. Figures are available from the author upon request.

effects over participation in only one classroom training in the second period and participation in one classroom training followed by a One-Euro-Job has mostly no beneficial effects over participation in a One-Euro-Job in the second period only. Moreover, immediately assigning individuals to classroom training is more effective than waiting and assigning them to classroom training in a second period (i.e., the effects of timing) by providing a work-test function. However, evidence of programme careers or stepwise integration is observed only for the sequence (CT,CT) versus the sequence (UBII,UBII). Because classroom training may target individuals with better labour market prospects, stepwise integration might not be necessary; however, programme careers might not arise for these individuals. Furthermore, the duration of classroom training and the considered periods are very short; therefore, individuals experience a smaller loss in human capital during their welfare receipt. In analysing longer duration of periods and sequences of One-Euro-Jobs that target hard-to-place individuals, Dengler (2013) finds that programme careers and stepwise integration do play a major role. Moreover, the effects for older-aged individuals aged at least 25 years are quite similar to the effects for the entire sample. Some differences are observed, but they are not significant.

Thus, some of the analysed sequences suggest a successful activation strategy to integrate individuals into regular employment. In particular, hard-to-place individuals who may be long-term unemployed and who may encounter extreme difficulty in finding jobs may require more assistance from job centres. Thus, intensified activation packages or sequences of ALMPs may be helpful for a successful integration into regular employment for such individuals, especially in times of high long-term unemployment. However, neither intensified activation packages nor strategic sequences of ALMPs are in place in many countries. Indeed, case workers often implement sequences very flexible: aspects of sequences such as the design of sequences, the upper limit of potential ALMPs per individual and the target groups are not regulated by legislation. However, targeting plays an even more important role in the case of sequences of ALMPs because negative effects that may arise from poorly targeting individuals into ALMPs may be strengthened. Moreover, programme careers may emerge if individuals participate in many ALMPs over several years without any positive employment effects. Many other countries have decentralised the activation of welfare recipients, leading to flexible activation schemes (Van Berkel 2010) and sequences may be also common in other countries, as studies on Switzerland (Lechner 2009), Austria (Lechner/Wiehler 2013), and Denmark (Graversen 2004) already have shown. Accordingly, flexible implementation of sequences may also arise in other countries. Thus, knowledge of the effectiveness of sequences or intensified activation packages is essential to implement successful strategic sequences or intensified activation packages that may integrate hard-toplace individuals into regular employment.

Several issues remain for future research. First, because this paper considers West Germany only, future research may analyse the effects of sequences of classroom training for East Germany. Second, longer sequences or sequences consisting of ALMPs other than those analysed in either this paper or that of Dengler (2013) would be interesting. However, the dynamic matching approach is very data hungry and very restrictive because of the definition of periods. Third, sanctions may be an important intermediate outcome, but data on sanctions for the considered period in this paper is not available before 2007. Thus, future research may analyse a later period to obtain information on sanctions. Fourth, sequences of ALMPs are highly effective for female participants in terms of employment outcomes, but female participants do not leave UB-II-receipt at the same time if they participate in the sequence of two classroom trainings compared with two periods of UB-II-receipt. Thus, future research may analyse this subgroup in a more detailed manner.

References

Achatz, Juliane; Eberle, Johanna; Jozwiak, Eva; Popp, Sandra; Schels, Brigitte (2009): Einsatz und Ausgestaltung von Trainingsmaßnahmen im Rechtskreis SGB II - die Perspektive von Vermittlungsfachkräften. IAB-Ergebnisbericht. Nuremberg.

Achatz, Juliane; Fehr, Sonja; Schels, Brigitte; Wolff, Joachim (2012): Wovon junge Arbeitslose im SGB II am meisten profitieren. IAB-Kurzbericht 06/2012. Nuremberg.

Bellmann, Lutz; Bender, Stefan; Bernhard, Stefan; Crimmann, Andreas; Dietrich, Hans; Dietz, Martin ; Haas, Anette ; Hirseland, Andreas; Hofmann, Barbara; Hohmeyer, Katrin; Koch, Susanne; König, Marion; Krug, Gerhard; Kruppe, Thomas; Kupka, Peter; Lietzmann, Torsten; Lott, Margit; Möller, Joachim; Plicht, Hannelore; Rauch, Angela; Rudolph, Helmut; Schreyer, Franziska; Sowa, Frank; Spitznagel, Eugen; Stephan, Gesine; Stops, Michael; Walwei, Ulrich; Wiemers, Jürgen; Wießner, Frank; Wolff, Joachim; Zabel, Cordula; Ziegler, Kerstin (2011): Neugestaltung der Förderinstrumente für Arbeitslose. IAB-Stellungnahme 09/2011. Nuremberg.

Bergemann, Annette; Van den Berg, Gerard J. (2008): Active Labor Market Policy Effects for Women in Europe - A Survey. In: Annales d'Économie et de Statistique, 91/92, p. 385–408.

Bernhard, Sarah; Kruppe, Thomas (2012): Effectiveness of Further Vocational Training in Germany – Empirical Findings for Persons Receiving Means-tested Unemployment Benefits. In: Schmollers Jahrbuch, 132(4), p. 501-526.

Biewen, Martin; Fitzenberger, Bernd; Osikominu, Aderonke; Waller, Marie (2007): Which Program for Whom? Evidence on the Comparative Effectiveness of Public Sponsored Training Programs in Germany. IZA Dicsussion Paper No. 2885. Bonn.

Büttner, Thomas (2008): Ankündigungseffekt oder Maßnahmewirkung? Eine Evaluation von Trainingsmaßnahmen zur Überprüfung der Verfügbarkeit. In: Zeitschrift für Arbeitsmarktforschung, 41(1), p. 25–40.

Caliendo, Marco; Kopeinig, Sabine (2008): Some Practical Guidance for the Implementation of Propensity Score Matching. In: Journal of Economic Surveys, 22(1), p. 31-72.

Caliendo, Marco; Mahlstedt, Robert; Mitnik, Oscar A. (2014): Unobservable, but Unimportant? The Influence of Personality Traits (and Other Usually Unobserved Variables) for the Evaluation of Labor Market Policies. IZA Dicsussion Paper No. 8337. Bonn.

Calmfors, Lars (1994): Active Labour Market Policy and Unemployment – A Framework for the Analysis of Crucial Design Features. In: OECD Economic Studies, 22, p. 7–47.

Card, David; Kluve, Jochen; Weber, Andrea (2010): Active Labour Market Policy Evaluations: A Meta-Analysis. In: The Economic Journal, 120(548), p. F452-F477.

Dahl, Espen (2003): Does 'workfare' work? The Norwegian experience. In: International Journal of Social Welfare, 12, p. 274-288.

Dengler, Katharina (2013): Effectiveness of sequences of One-Euro-Jobs. Is it better to do more One-Euro-Jobs or to wait? IAB-Discussion Paper 16/2013. Nuremberg.

Dengler, Katharina; Hohmeyer, Katrin (2010): Maßnahmesequenzen im SGB II. Eine deskriptive Analyse. IAB-Forschungsbericht 08/2010. Nuremberg.

Dengler, Katharina; Hohmeyer, Katrin; Moczall, Andreas; Wolff, Joachim (2013): Intensified activation for disadvantaged welfare recipients in Germany: Does it work? In: The International Journal of Sociology and Social Policy, 33(5/6), p. 289– 308.

Department for Statistics of the Federal Employment Agency (2006-2009): Leistungen zur Eingliederung an erwerbsfähige Hilfebedürftige: Einsatz von Arbeitsgelegenheiten 2005/2006/2007/2008, Nuremberg.

Department for Statistics of the Federal Employment Agency (2013): Arbeitslosigkeit im Zeitverlauf, Nuremberg.

Department for Statistics of the Federal Employment Agency (2014): Data Warehouse, Nuremberg.

Finn, Daniel; Schulte, Bernd (2008): 'Employment First': Activating the British Welfare State. In: Eichhorst, Werner; Kaufmann, Otto; Konle-Seidl, Regina (Eds.) (2008): Bringing the Jobless into Work? Experiences with Activation Schemes in Europe and the US, Berlin Heidelberg: Springer, p. 297-343.

Fitzenberger, Bernd; Orlanski, Olga; Osikominu, Aderonke; Paul, Marie (2013): Déjà Vu? Short-term trainings in Germany 1980-1992 and 2000-2003. In: Empirical Economics, 44(1), p. 289-328.

Frölich, Markus (2004): Programme evaluation with multiple treatments. In: Journal of Economic Surveys, 18(2), p. 181–224.

Gerfin, Michael; Lechner, Michael (2002): A Microeconometric Evaluation of the Active Labour Market Policy in Switzerland. In: The Economic Journal, 112(482), p. 854–893.

Graversen, Brian Krogh (2004): Employment effects of active labour market programmes: Do the programmes help welfare benefit recipients to find jobs? Aarhus: University of Aarhus.

Graversen, Brian Krogh; van Ours, Jan C. (2008): How to help unemployed find jobs quickly: Experimental evidence from a mandatory activation program. In: Journal of Public Economics, 92(10-11), p. 2020–2035.

Greenberg, David H.; Michalopoulos, Charles; Robins, Philip K. (2003): A Meta-Analysis of Government-Sponsored Training Programs. In: Industrial and Labor Relations Review, 57(1), p. 31–53.

Hagen, Tobias; Steiner, Viktor (2000): Von der Finanzierung der Arbeitslosigkeit zur Förderung von Arbeit. Analysen und Empfehlungen zur Arbeitsmarktpolitik in Deutschland, Baden-Baden: Nomos Verlagsgesellschaft.

Hartig, Martina; Jozwiak, Eva; Wolff, Joachim (2008): Trainingsmaßnahmen: Für welche unter 25-jährigen Arbeitslosengeld-II-Empfänger erhöhen sie die Beschäftigungschancen? IAB-Forschungsbericht 06/2008. Nuremberg.

Hohmeyer, Katrin (2012): Effectiveness of One-Euro-Jobs: Do programme characteristics matter? In: Applied Economics, 44(34), p. 4469–4484.

Hohmeyer, Katrin; Wolff, Joachim (2012): A fistful of euros: Is the German one-euro job workfare scheme effective for participants? In: International Journal of Social Welfare, 21(2), p. 174–185.

Hohmeyer, Katrin; Wolff, Joachim (2010): Direct job creation revisited. Is it effective for welfare recipients and does it matter whether participants receive a wage? IAB-Discussion Paper 21/2010. Nuremberg.

Huber, Martin; Lechner, Michael; Wunsch, Conny; Walter, Thomas (2011): Do German Welfare-to-Work Programmes Reduce Welfare Dependency and Increase Employment? In: German Economic Review, 12(2), p. 182–204.

Hujer, Reinhard; Thomsen, Stephan L.; Zeiss, Christopher (2006): The Effects of Short-Term Training Measures on the Individual Unemployment Duration in West Germany. ZEW Discussion Paper No. 06-065. Mannheim.

Immervoll, Herwig; Scarpetta, Stefano (2012): Activation and employment support policies in OECD countries. An overview of current approaches. In: IZA Journal of Labor Policy, 1(9).

Jacobi, Lena; Kluve, Jochen (2007): Before and After the Hartz Reforms: The Performance of Active Labour Market Policy in Germany. In: Zeitschrift für Arbeitsmarktforschung, 40(1), p. 45–64.

Jaenichen, Ursula; Stephan, Gesine (2011): The effectiveness of targeted wage subsidies for hard-to-place workers. In: Applied Economics, 43(10), p. 1209–1225.

Kluve, Jochen (2010): The effectiveness of European active labor market programs. In: Labour Economics, 17(6), p. 904-918.

Kopf, Eva (2013): Short training for welfare recipients in Germany: which types work? In: International Journal of Manpower, 34(5), p. 486-516.

Lechner, Michael (1999): Earnings and Employment Effects of Continuous Off-the-Job Training in East Germany after Unification. In: Journal of Business & Economics Statistics, 17(1), p. 74–90.

Lechner, Michael (2004): Sequential Matching Estimation of Dynamic Causal Models. Universität St. Gallen Discussion paper 2004-06. St. Gallen.

Lechner, Michael (2008): Matching Estimation of Dynamic Treatment Models: Some Practical Issues. In: Millimet, Daniel L.; Smith, Jeffrey A.; Vyctlacil, Edward (Eds.) (2008): Advances in Econometrics, Volume 21, Modelling and Evaluating Treatment Effects in Econometrics, Amsterdam: Elsevier, p. 289–333.

Lechner, Michael (2009): Sequential Causal Models for the Evaluation of Labor Market Programs. In: Journal of Business & Economic Statistics, 27(1), p. 71–83.

Lechner, Michael; Miquel, Ruth (2010): Identification of the effects of dynamic treatments by sequential conditional independence assumptions. In: Empirical Economics, 39(1), p. 111–137.

Lechner, Michael; Miquel, Ruth; Wunsch, Conny (2011): Long-run Effects of Public Sector Sponsored Training in West Germany. In: Journal of the European Economic Association, 9(4), p. 742–784.

Lechner, Michael; Wiehler, Stephan (2013): Does the Order and Timing of Active Labour Market Programmes Matter? In: Oxford Bulletin of Economics and Statistics, 75(2), p. 180–212.

Lissenburgh, Stephen (2004): New deal option effects on employment entry and unemployment exit. An evaluation using propensity score matching. In: International Journal of Manpower, 25(5), p. 411-330.

OECD (2013): Long-term unemployment. OECD Factbook 2013: Economic, Environmental and Social Statistics. Paris.

OECD Statistics (2014): OECD Employment and Labour Market Statistics (database): Unemployment by duration: incidence, Paris.

Osikominu, Aderonke (2013): Quick Job Entry or Long-Term Human Capital Development? The Dynamic Effects of Alternative Training Schemes. In: Review of Economic Studies, 80(1), p. 313-342.

Pedersen, Jonas Maibom; Rosholm, Michael; Svarer, Michael (2012): Experimental Evidence on the Effects of Early Meetings and Activation. IZA Dicsussion Paper No. 6970. Bonn.

Pissarides, Christopher A. (1979): Job Matchings with State Employment Agencies and Random Search. In: The Economic Journal, 89(356), p. 818–833.

Robins, James (1986): A new approach to causal inference in mortality studies with a sustained exposure period-application to control of the healthy worker survivor effect. In: Mathematical Modelling, 7, p. 1393–1512.

Romeu Gordo, Laura; Wolff, Joachim (2011): Creating Employmen or Keeping Them Busy? An Evaluation of Training Programes for Older Workers in Germany. In: Journal of Aging and Social Policy, 23(2), p. 198-218.

Rønsen, Marit; Skarðhamar, Torbjørn (2009): Do welfare-to-work initiatives work? Evidence from an activation programme targeted at social assistance recipients in Norway. In: Journal of European Social Policy, 19(1), p. 61-77.

Rosenbaum, Paul R.; Rubin, Donald B. (1983): The Central Role of the Propensity Score in Observational Studies for Causal Effects. In: Biometrika, 70(1), p. 41–55.

Rosholm, Michael (2008): Experimental Evidence on the Nature of the Danish Employment Miracle. IZA Dicsussion Paper No. 3620. Bonn.

Roy, Andrew D. (1951): Some Thoughts on The Distribution of Earnings. In: Oxford Economic Papers, 3(2), p. 135–146.

Rubin, Donald B. (1980): Randomization Analysis of Experimental Data: The Fisher Randomization Test Comment. In: Journal of the American Statistical Association, 75(371), p. 591–593.

Rubin, Donald B. (1974): Estimating Causal Effects of Treatments in Randomized and Nonrandomized Studies. In: Journal of Educational Psychology, 66(5), p. 688–701.

Sianesi, Barbara (2004): An Evaluation of the Swedish System of Active Labor Market Programs in the 1990s. In: The Review of Economics and Statistics, 86(1), p. 133–155.

Steinke, Joß; Koch, Susanne; Kupka, Peter; Osiander, Christoph; Dony, Elke; Güttler, Deflef; Hesse, Claudia; Knapp, Barbara (2012): Neuorientierung der Arbeitsmarktpolitik. Die Neuausrichtung der arbeitsmarktpolitischen Instrumente aus dem Jahr 2009 im Blickpunkt: Mehr Flexibilität und größere Handlungsspielräume für die Vermittler? IAB-Forschungsbericht 02/2012. Nuremberg.

Stephan, Gesine (2008): The Effects of Active Labor Market Programs in Germany: An Investigation Using Different Definitions of Non-Treatment. In: Jahrbücher für Nationalökonomie und Statistik, 228(5+6), p. 586-611.

Stephan, Gesine; Pahnke, André (2011): The relative effectiveness of selected active labor market programs: An empirical investigation for Germany. In: The Manchester School, 79(6), p. 1262-1293.

Stephan, Gesine; Rässler, Susanne; Schewe, Torben (2006): Das TrEffeR-Projekt der Bundesagentur für Arbeit: Die Wirkung von Maßnahmen aktiver Arbeitsmarktpolitik. In: Zeitschrift für Arbeitsmarktforschung, 39(3/4), p. 447-465.

Thomsen, Stephan L.; Walter, Thomas (2010): Temporary Extra Jobs for Immigrants: Merging Lane to Employment or Dead-End Road in Welfare? In: Labour, 24(s1), p. 114–140.

Thomsen, Stephan L.; Walter, Thomas; Aldashev, Alisher (2013): Short-term training programs for immigrants in the German welfare system: do effects differ from natives and why? In: IZA Journal of Migration, 2(24), p. 1-20.

Van Berkel, Rik (2010): The Provision of Income Protection and Activation Services for the Unemployed in 'Active' Welfare States. An International Comparison. In: Journal of Social Policy, 39(1), p. 17-34.

Wolff, Joachim; Jozwiak, Eva (2007): Does short-term training activate meanstested unemployment benefit recipients in Germany? IAB-Discussion Paper 29/2007. Nuremberg.

Wolff, Joachim; Popp, Sandra; Zabel, Cordula (2010): Ein-Euro-Jobs für hilfebedürftige Jugendliche. Hohe Verbreitung, geringe Integrationswirkung. In: WSI-Mitteilungen, 63(1), p. 11–18.

Wunsch, Conny; Lechner, Michael (2008): What did all the money do? On the general ineffectiveness of recent West German Labour Market Programmes. In: Kyklos, 61(1), p. 134-174.

Yankova, Katerina (2010): Der Selektionsprozess in Maßnahmen der aktiven Arbeitsmarktpolitik. Eine explorative Untersuchung für die deutsche Arbeitsvermittlung. IAB-Discussion Paper 11/2010. Nuremberg.

Zabel, Cordula (2013): Does activation increase lone mothers' employment chances? In: International Journal of Sociology and Social Policy, 33(7/8), p. 453-473.

Figures and Tables

Figures

Figure 1





Source: OECD Statistics (2014).

Figure 2 Definition of states and sequences



Source: Own Illustration.

Figure 3 Dynamic window approach



Source: Own Illustration.

Figure 4





Source: Own Calculations.

Figure 5 DATET for (CT,CT) versus (UBII,UBII) for participants in classroom training in the first period (CT) on no UB-II-receipt



Source: Own Calculations.

Figure 6

DATET for (CT,CT) versus (UBII,UBII) for participants in classroom training in the first period (CT) on One-Euro-Jobs



Source: Own Calculations.

Figure 7 DATET for (CT,CT) versus (UBII,UBII) for participants in classroom training in the first period (CT) on short-term training



Source: Own Calculations.

Figure 8

DATET for (CT,CT) versus (UBII,UBII) for participants in classroom training in the first period (CT) on further vocational training



Source: Own Calculations.

Figure 9 DATET for (CT,UBII) versus (UBII,CT) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Figure 10

DATET for (CT,UBII) versus (UBII,CT) for participants in classroom training in the first period (CT) on no UB-II-receipt



Source: Own Calculations.

Figure 11 DATET for (CT,UBII) versus (UBII,CT) for participants in classroom training in the first period (CT) on One-Euro-Jobs



Source: Own Calculations.





Source: Own Calculations.

Figure 13 DATET for (CT,UBII) versus (UBII,CT) for participants in classroom training in the first period (CT) on further vocational training



Source: Own Calculations.

Figure 14

DATET for (CT,CT) versus (UBII,CT) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Figure 15 DATET for (CT,CT) versus (UBII,CT) for participants in classroom training in the first period (CT) on no UB-II-receipt



Source: Own Calculations.

Figure 16

DATET for (CT,CT) versus (UBII,CT) for participants in classroom training in the first period (CT) on One-Euro-Jobs



Source: Own Calculations.

Figure 17 DATET for (CT,CT) versus (UBII,CT) for participants in classroom training in the first period (CT) on short-term training



Source: Own Calculations.

Figure 18

DATET for (CT,CT) versus (UBII,CT) for participants in classroom training in the first period (CT) on further vocational training



Source: Own Calculations.

Figure 19 DATET for (CT,1EJ) versus (UBII,1EJ) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Figure 20

DATET for (CT,1EJ) versus (UBII,1EJ) for participants in classroom training in the first period (CT) on no UB-II-receipt



Source: Own Calculations.

Figure 21 DATET for (CT,1EJ) versus (UBII,1EJ) for participants in classroom training in the first period (CT) on One-Euro-Jobs



Source: Own Calculations.

Figure 22

DATET for (CT,1EJ) versus (UBII,1EJ) for participants in classroom training in the first period (CT) on short-term training



Source: Own Calculations.

Figure 23 DATET for (CT,1EJ) versus (UBII,1EJ) for participants in classroom training in the first period (CT) on further vocational training



Source: Own Calculations.

Figure 24

All vs. older-aged West German men – DATET for (CT,CT) versus (UBII,UBII) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Figure 25 All vs. older-aged West German women – DATET for (CT,CT) versus (UBII,UBII) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Figure 26

All vs. older-aged West German men – DATET for (CT,UBII) versus (UBII,CT) for participants in classroom training in the first period (CT) on the regular employment rate



Figure 27 All vs. older-aged West German women – DATET for (CT,UBII) versus (UBII,CT) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Figure 28

All vs. older-aged West German men – DATET for (CT,CT) versus (UBII,CT) for participants in classroom training in the first period (CT) on the regular employment rate



Figure 29

All vs. older-aged West German women – DATET for (CT,CT) versus (UBII,CT) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Figure 30

All vs. older-aged West German men – DATET for (CT,1EJ) versus (UBII,1EJ) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Figure 31 All vs. older-aged West German women – DATET for (CT,1EJ) versus (UBII,1EJ) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Figure 32

Robustness check (caliper) – DATET for (CT,CT) versus (UBII,UBII) for participants in classroom training in the first period (CT) on the regular employment rate



Figure 33

Robustness check (caliper) – DATET for (CT,UBII) versus (UBII,CT) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Figure 34

Robustness check (caliper) – DATET for (CT,CT) versus (UBII,CT) for participants in classroom training in the first period (CT) on the regular employment rate



Figure 35

Robustness check (caliper) – DATET for (CT,1EJ) versus (UBII,1EJ) for participants in classroom training in the first period (CT) on the regular employment rate



Source: Own Calculations.

Tables

Table 1

Average stock of unemployed UB-II-recipients and inflow into major ALMPs between 2005 and 2008 for West German men and women (in 1,000)

	2005		2006		2007		2008	
	Men	Women	Men	Women	Men	Women	Men	Women
Average stock of unemployed UB-II-recipients	882	686	872	725	737	666	651	614
Classroom training	103	72	103	77	105	85	115	93
In-firm training	55	23	74	32	80	37	73	37
Further vocational training	24	15	41	25	52	33	66	45
One-Euro-Jobs	207	108	264	142	255	148	236	147
Work opportunity as contributory employment	6	3	12	6	13	6	19	11
Wage subsidy for employers	26	9	50	17	58	22	51	23
Wage subsidy for employees	0.4	0.3	4	2	7	4	8	4
Start-up subsidy	8	3	14	6	11	5	7	4

Note: Data without information from local authorities.

Source: Department for Statistics of the German Federal Employment Agency (2014).

Table 2Sample sizes and selected descriptive statistics

	San	nple	С	т	U	BII	(СТ,	СТ)	(CT,U	JBII)	(UBI	I,CT)	(UBII,	UBII)	(CT,	1EJ)	(UBII,	1EJ)
	West G	ermany	West G	ermany	West G	ermany	West Ge	ermany	West Ge	ermany	West G	ermany	West G	ermany	West G	ermany	West Ge	ərmany
	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women	Men	Women
Observations	350,786	364,294	11,588	7,754	269,930	306,307	554	420	6,157	4,609	3,267	2,753	201,631	250,052	710	380	5,494	3,242
Variables at/before entry																		
Average age at entry	32.9	32.3	31.5	32.4	33.3	32.4	30.5	31.7	32.6	33.5	32.3	33.3	33.7	32.5	28.9	30.6	32.5	32.1
Children at entry (in %)																		
no	77.6	51.1	82.2	68.9	77.0	48.9	86.8	70.5	81.1	65.6	80.5	59.9	76.2	47.1	88.0	77.9	85.0	68.2
yes	22.4	48.9	17.8	31.1	23.0	51.1	13.2	29.5	18.9	34.4	19.5	40.1	23.8	52.9	12.0	22.1	15.0	31.8
Average cum. duration of minor employment 5 years																	~~ ~	
before entry (in days)	113.3	244.0	111.7	226.7	112.8	243.7	123.3	203.0	101.6	231.4	110.7	247.2	110.1	238.7	95.6	167.1	82.9	1/8.4
Average cum. duration of UB II 1 year before entry (in days)	34.9	24.6	33.9	25.5	33.7	23.9	34.2	25.0	31.3	23.9	34.0	25.8	31.6	22.7	36.7	27.7	40.5	28.7
Any last contributory employment before entry (in %)																		
yes	77.5	64.0	79.5	69.5	76.1	62.6	76.2	69.8	78.3	66.7	81.3	71.8	74.2	60.9	73.0	68.9	78.2	65.3
no	22.5	36.0	20.5	30.5	23.9	37.4	23.8	30.2	21.7	33.3	18.7	28.2	25.8	39.1	27.0	31.1	21.8	34.7
Average equivalent household income from welfare in the month at entry (in Euros)	381.3	301.1	394 3	351 3	381 7	297 0	397 9	343.4	397.8	346 3	404 9	338.8	382.3	296 7	392.3	398.4	421.8	377 5
Intermediate variables (before period 2)	001.0	001.1	004.0	001.0	001.7	201.0	007.0	040.4	007.0	040.0	404.0	000.0	002.0	200.7	002.0	000.4	421.0	077.0
Intermediate: Average cum. duration of minor employment																		
1 month before period 2 (in days)							1.7	2.9	2.1	3.8	2.7	4.3	3.3	4.9	1.2	2.2	1.6	2.8
Intermediate: Average cum. duration of UB II 1 month																		
before period 2 (in days)							30.3	30.4	30.3	30.3	30.4	30.4	30.3	30.4	30.3	30.3	30.3	30.4
Intermediate: Minor employment at period 2 (in %)																		
no							94.4	89.8	92.6	86.9	90.9	85.4	88.9	83.9	96.2	93.4	94.6	91.1
yes							5.6	10.2	7.4	13.1	9.1	14.6	11.1	16.1	3.8	6.6	5.4	8.9
Intermediate: Children at period 2 (in %)																		
no							86.5	70.0	80.6	65.7	79.6	59.8	75.5	45.4	86.8	77.6	84.2	68.0
yes							13.5	30.0	19.4	34.3	20.4	40.2	24.5	54.6	13.2	22.4	15.8	32.0
Intermediate: Average equivalent household income from																		
welfare in the month before period 2 (in Euros)							579.6	504.1	586.8	495.9	568.0	482.4	533.1	401.2	570.3	523.9	592.9	516.2

Table 3			
Outcomes 12 months after	programme start in	the second pe	eriod

	Men - West Germany				Wo	Women - West Germany			
Outcomes 12 months after									
programme start in the second	All	All	Matched	Matched	All	All	Matched	Matched	
period (in %) for sequence k	controls	treated	controls	treated	controls	treated	controls	treated	
(treated) vs. sequence h (controls)		lioutou	00111.010	lioalou			00111010		
	1								
Regular employment rate	19.4	26.7	22.1	26.9	11.4	22.4	13.1	24.2	
Short-term training	0.6	0.9	0.5	1.0	0.4	0.7	0.4	0.3	
Further vocational training	0.9	3.8	0.8	3.4	0.5	2.9	0.6	2.8	
One-Euro-Jobs	3.0	5.6	3.2	5.5	1.8	5.7	1.9	5.6	
No UB-II-receipt	36.6	42.4	40.0	44.4	29.1	30.7	30.6	31.4	
(CT,UBII) vs. (UBII,CT)									
Regular employment rate	25.3	22.2	25.6	22.7	19.1	16.8	19.7	17.6	
Short-term training	0.9	0.8	0.8	0.8	1.1	0.8	1.0	0.9	
Further vocational training	2.6	1.6	2.9	1.7	2.3	1.9	2.3	2.0	
One-Euro-Jobs	5.0	4.4	4.8	4.0	4.8	4.3	4.6	4.1	
No UB-II-receipt	35.7	37.9	37.0	39.0	30.6	31.1	31.2	32.3	
(CT,CT) vs. (UBII,CT)									
Regular employment rate	25.3	26.7	25.6	26.9	19.1	22.4	19.7	24.2	
Short-term training	0.9	0.9	0.8	1.0	1.1	0.7	1.0	0.3	
Further vocational training	2.6	3.8	2.9	3.4	2.3	2.9	2.3	2.8	
One-Euro-Jobs	5.0	5.6	4.8	5.5	4.8	5.7	4.6	5.6	
No UB-II-receipt	35.7	42.4	37.0	44.4	30.6	30.7	31.2	31.4	
(CT,1EJ) vs. (UBII,1EJ)									
Regular employment rate	17.9	19.6	18.5	20.1	14.8	15.0	15.4	16.1	
Short-term training	0.5	0.7	0.4	1.0	0.9	0.8	0.8	1.1	
Further vocational training	1.3	2.3	1.3	2.3	1.3	3.2	1.5	3.6	
One-Euro-Jobs	13.3	12.1	13.4	11.4	12.8	15.5	13.4	15.4	
No UB-II-receipt	30.6	34.4	31.5	36.6	26.0	25.8	26.3	25.7	

Table 4Mean standardised absolute bias

	Men - Wes	t Germany	Women - Wo	est Germany
Mean standarised absolute bias (MSB)	Before matching	After matching	Before matching	After matching
(CT,CT) vs. (UBII,UBII) for CT				
Dynamic matching 1: UBII to CT	8.29	0.76	13.36	1.39
Dynamic matching 2: (UBII,UBII) to CT	10.20	0.97	16.06	1.04
Dynamic matching 3: (CT,CT) to CT	7.58	2.88	4.33	2.04
Final matching: (CT,CT) to (UBII,UBII)	12.21	5.77	19.39	11.90
(CT,UBII) vs. (UBII,CT) for CT			-	
Dynamic matching 1: UBII to CT	8.29	0.76	13.36	1.39
Dynamic matching 2: (UBII,CT) to CT	5.58	1.05	7.44	1.03
Dynamic matching 3: (CT,UBII) to CT	8.00	1.76	9.71	2.03
Final matching: (CT,UBII) to (UBII,CT)	5.33	1.92	4.85	4.25
(CT,CT) vs. (UBII,CT) for CT				
Dynamic matching 1: UBII to CT	8.29	0.76	13.36	1.39
Dynamic matching 2: (UBII,CT) to CT	5.58	1.05	7.44	1.03
Dynamic matching 3: (CT,CT) to CT	7.58	2.88	4.33	2.04
Final matching: (CT,CT) vs. (UBII,CT)	8.05	4.37	9.27	5.08
(CT,1EJ) vs. (UBII,1EJ) for CT	-		-	
Dynamic matching 1: UBII to CT	8.29	0.76	13.36	1.39
Dynamic matching 2: (UBII,1EJ) to CT	8.33	1.60	6.85	2.05
Dynamic matching 3: (CT,1EJ) to CT	11.21	2.94	10.79	4.54
Final matching: (CT,1EJ) to (UBII,1EJ)	10.31	4.79	8.57	6.01

Table 5Individuals aged at least 25 years – mean standardised absolute bias

	Men - Wes	st Germany	Women - Wo	est Germany
Mean standarised absolute bias (MSB)	Before matching	After matching	Before matching	After matching
(CT,CT) vs. (UBII,UBII) for CT				
Dynamic matching 1: UBII to CT	6.94	0.93	14.72	1.17
Dynamic matching 2: (UBII,UBII) to CT	9.23	0.87	16.05	1.35
Dynamic matching 3: (CT,CT) to CT	6.50	2.87	4.42	4.19
Final matching: (CT,CT) to (UBII,UBII)	9.39	5.79	19.88	13.06
(CT,UBII) vs. (UBII,CT) for CT				
Dynamic matching 1: UBII to CT	6.94	0.93	14.72	1.17
Dynamic matching 2: (UBII,CT) to CT	4.93	1.30	5.17	1.80
Dynamic matching 3: (CT,UBII) to CT	7.92	2.19	10.72	2.17
Final matching: (CT,UBII) to (UBII,CT)	6.86	1.76	5.15	3.01
(CT,CT) vs. (UBII,CT) for CT				
Dynamic matching 1: UBII to CT	6.94	0.93	14.72	1.17
Dynamic matching 2: (UBII,CT) to CT	4.93	1.30	5.17	1.80
Dynamic matching 3: (CT,CT) to CT	6.50	2.87	4.42	4.19
Final matching: (CT,CT) vs. (UBII,CT)	7.03	4.53	7.04	6.22
(CT,1EJ) vs. (UBII,1EJ) for CT	-		-	
Dynamic matching 1: UBII to CT	6.94	0.93	14.72	1.17
Dynamic matching 2: (UBII,1EJ) to CT	11.74	1.97	8.73	2.41
Dynamic matching 3: (CT,1EJ) to CT	11.84	4.88	11.49	7.00
Final matching: (CT,1EJ) to (UBII,1EJ)	7.48	5.78	7.87	8.11
Table 6 Robustness Check (Caliper Matching) – mean standardised absolute bias

	Men - Wes	st Germany	Women - West Germany	
Mean standarised absolute bias (MSB)	Before matching	After matching	Before matching	After matching
(CT,CT) vs. (UBII,UBII) for CT				
Dynamic matching 1: UBII to CT	8.29	0.86	13.36	1.44
Dynamic matching 2: (UBII,UBII) to CT	8.86	0.97	14.72	1.27
Dynamic matching 3: (CT,CT) to CT	7.58	2.61	4.33	1.94
Final matching: (CT,CT) to (UBII,UBII)	12.21	7.38	19.39	13.24
(CT,UBII) vs. (UBII,CT) for CT				
Dynamic matching 1: UBII to CT	8.29	0.86	13.36	1.44
Dynamic matching 2: (UBII,CT) to CT	4.68	0.95	6.42	1.05
Dynamic matching 3: (CT,UBII) to CT	7.99	1.55	9.71	2.18
Final matching: (CT,UBII) to (UBII,CT)	5.33	2.91	4.85	4.65
(CT,CT) vs. (UBII,CT) for CT				
Dynamic matching 1: UBII to CT	8.29	0.86	13.36	1.44
Dynamic matching 2: (UBII,CT) to CT	4.68	0.95	6.42	1.05
Dynamic matching 3: (CT,CT) to CT	7.49	3.09	4.33	1.94
Final matching: (CT,CT) vs. (UBII,CT)	8.05	5.13	9.27	5.86
(CT,1EJ) vs. (UBII,1EJ) for CT				
Dynamic matching 1: UBII to CT	8.29	0.86	13.36	1.44
Dynamic matching 2: (UBII,1EJ) to CT	8.08	1.70	6.71	1.75
Dynamic matching 3: (CT,1EJ) to CT	11.21	2.94	10.79	4.38
Final matching: (CT,1EJ) to (UBII,1EJ)	10.31	5.41	8.57	6.09

Source: Own Calculations.

Appendix

Table A-1 Short matching protocol according to Lechner (2004): A. Definition of the sequences (\underline{s}_2^h and \underline{s}_2^k) and the population s_1^k **B.** Matching of $\underline{s}_2^h = (s_1^h, s_2^h)$ to s_1^k 1. Definition of weight $w_i^{\frac{s^2}{2}} = 0$ for units of \underline{s}_2^h 2. Estimation of probit $P(S_1 = s_1^h | \underline{X}_o = x_o) = p_1^{s_1^h}$ 3. Common Support 4. Matching of s_1^k to s_1^h that is closest in terms of $p^{s_1^h}$ using nearest neighbour matching (one-to-one) with replacement 5. Estimation of probit $P(S_2 = s_2^h | S_1 = s_1^h, X_1 = x_1) = p^{s_2^h | s_1^h}$ 6. Common Support 7. Matching of s_1^k to \underline{s}_2^h that is closest in terms of $p^{s_1^h}$ and $p^{s_2^h|s_1^h}$ using the Mahalanobis metric 8. Increase of weight $w_i^{\underline{s}_2^h}$ by 1 every time an observation in \underline{s}_2^h is matched C. Matching of $s_2^k = (s_1^k, s_2^k)$ to s_1^k 1. Definition of weight $w_i^{\frac{S^k}{2}} = 0$ for units of \underline{s}_2^k 2. Estimation of probit $P(S_2 = s_2^k | S_1 = s_{1,X_1}^k = x_1) = p^{s_2^k | s_1^k}$ 3. Common Support 4. Matching of s_1^k to \underline{s}_2^k that is closest in term of $p^{s_2^k|s_1^k}$ using nearest neighbour matching (one-to-one) with replacement 5. Increase of weight $w_i^{\underline{s}_2^k}$ by 1 every time an observation in \underline{s}_2^k is matched **D. Joint Common Support** 1. Reduction of $w_i^{\frac{S^k}{2}}$ by 1 for every observation i matched to s_1^k , but deleted in B.3 or B.6 2. Reduction of $w_i^{\frac{S_2^h}{2}}$ by 1 for every observation i matched to s_1^k , but deleted in C.3

E. Estimation of DATET and variance

1. Estimation of DATET:

$$\hat{\theta}_{2}^{\underline{s}_{2}^{k},\underline{s}_{2}^{h}}(s_{1}^{k}) = \frac{1}{\sum_{i \in \underline{s}_{2}^{k}} w_{i}^{\underline{s}_{2}^{k}}} \sum_{i \in \underline{s}_{2}^{k}} w_{i}^{\underline{s}_{2}^{k}} y_{i} - \frac{1}{\sum_{i \in \underline{s}_{2}^{h}} w_{i}^{\underline{s}_{2}^{h}}} \sum_{i \in \underline{s}_{2}^{h}} w_{i}^{\underline{s}_{2}^{h}} y_{i}$$

2. Estimation of variance:

$$\widehat{Var}\left(\widehat{\theta}_{2}^{\underline{s}_{2}^{k},\underline{s}_{2}^{h}}(s_{1}^{k})\right) = \frac{\sum_{i \in \underline{s}_{2}^{k}}(w_{i}^{\underline{s}_{2}^{k}})^{2} \ \widehat{Var}(Y_{2}|S = \underline{s}_{2}^{k})}{(\sum_{i \in \underline{s}_{2}^{k}}w_{i}^{\underline{s}_{2}^{k}})^{2}} + \frac{\sum_{i \in \underline{s}_{2}^{h}}(w_{i}^{\underline{s}_{2}^{h}})^{2} \ \widehat{Var}(Y_{2}|S = \underline{s}_{2}^{h})}{(\sum_{i \in \underline{s}_{2}^{h}}w_{i}^{\underline{s}_{2}^{h}})^{2}}$$

with

$$\widehat{Var}(Y_2|S=\underline{s}_2) = \frac{1}{N^{\underline{s}_2}} \sum_{i \in \underline{s}_2} (y_i - \overline{y}_2^{\underline{s}_2})^2, \ \overline{y}_2^{\underline{s}_2} = \frac{1}{N^{\underline{s}_2}} \sum_{i \in \underline{s}_2} y_{2,i}, \ N^{\underline{s}_2} = \sum_i \underline{1} (\underline{s}_{2,i} = \underline{s}_2)$$

Recently published

No.	Author(s)	Title	Date
<u>9/2016</u>	Möller, J.	Lohnungleichheit - Gibt es eine Trendwende?	3/16
<u>10/2016</u>	Bossler, M. Gerner, HD.	Employment effects of the new German mini- mum wage: Evidence from establishment-level micro data	3/16
<u>11/2016</u>	Bossler, M. Grunau, P.	Asymmetric information in external versus inter- nal promotions	4/16
<u>12/2016</u>	Mendolicchio, C. Pietra, T.	A re-examination of constrained Pareto ineffi- ciency in economies with incomplete markets	4/16
<u>13/2016</u>	Hamann, S. Niebuhr, A. Peters, C.	Benefits of dense labour markets: Evidence from transitions to employment in Germany	4/16
<u>14/2016</u>	Bender, S. Bloom, N. Card, D. Van Reenen, J. Wolter, S.	Management practices, workforce selection, and productivity	5/16
<u>15/2016</u>	Bossler, M. Broszeit, S.	Do minimum wages increase job satisfaction? Micro data evidence from the new German min- imum wage	5/16
<u>16/2016</u>	Dengler, K. Stops, M. Vicari, B.	Occupation-specific matching efficiency	5/16
<u>17/2016</u>	Schierholz, M. Gensicke, M. Tschersich, N.	Occupation coding during the interview	5/16
<u>18/2016</u>	Lietzmann, T. Schmelzer, P. Wiemers, J.	Does marginal employment promote regular employment for unemployed welfare benefit re- cipients in Germany?	6/16
<u>19/2016</u>	Card, D. Cardoso, A. Heining, J. Kline, P.	Firms and labor market inequality	6/16
<u>20/2016</u>	Weber, E. Weigand, R.	Identifying macroeconomic effects of refugee migration to Germany	6/16
<u>21/2016</u>	Neffke, F. Otto, A. Weyh, A.	Inter-industry labor flows	6/16
<u>22/2016</u>	Blien, U. Ludewig, O.	Technological progress and (un)employment development	7/16
<u>23/2016</u>	Stepanok, I.	A North-South Model of Trade with Search Un- employment	7/16

As per: 2016-08-09

For a full list, consult the IAB website <u>http://www.iab.de/de/publikationen/discussionpaper.aspx</u>

Imprint

IAB-Discussion Paper 24/2016 12 August 2016

Editorial address Institute for Employment Research of the Federal Employment Agency Regensburger Str. 104 D-90478 Nuremberg

Editorial staff Ricardo Martinez Moya, Jutta Palm-Nowak

Technical completion Renate Martin

All rights reserved Reproduction and distribution in any form, also in parts, requires the permission of IAB Nuremberg

Website http://www.iab.de

Download of this Discussion Paper http://doku.iab.de/discussionpapers/2016/dp2416.pdf

ISSN 2195-2663

For further inquiries contact the author:

Katharina Dengler Phone +49.911.179 2941 E-mail Katharina.Dengler@iab.de